COLLECTIVE AGREEMENT

between

CITY OF TERRACE

and

CANADIAN UNION OF PUBLIC EMPLOYEES LOCAL 2012

January 1, 2022 - December 31, 2026

Table of Contents

ARTICLE 1	DEFINITIONS	8
1.01	APPROVED ABSENCE	8
1.02	CALLOUT	8
1.03	CHARGEHAND	8
1.04	CONTINUOUS SERVICE	8
1.05	EMPLOYEE	9
1.06	REGULAR FULL-TIME EMPLOYEE	9
1.07	REGULAR PART-TIME EMPLOYEE	9
1.08	AUXILIARY LIFEGUARD/INSTRUCTOR	9
1.09	TEMPORARY AND SEASONAL EMPLOYEES	9
1.10	CASUAL EMPLOYEE	9
1.11	STUDENT EMPLOYEE	10
1.12	Layoff	10
1.13	OVERTIME	10
1.14	OVERTIME CREDITS	
1.15	Personal Harassment	
1.16	Probation	
1.17	Definitions - Seniority	
1.18	Sexual Harassment	
1.19	SUNDAY EQUIVALENT	
1.20	TECHNOLOGICAL CHANGE	
4 0 4		- 11
1.21	WEEK	
1.21 1.22	WEEK	
		11
1.22	Spouse	11
1.22 ARTICLE 2 2.01	SPOUSE	11 12
1.22 ARTICLE 2	SPOUSE	111212
1.22 ARTICLE 2 2.01 ARTICLE 3	MANAGEMENT'S RIGHTS MANAGEMENT'S RIGHTS DISCRIMINATION No DISCRIMINATION	11121212
1.22 ARTICLE 2 2.01 ARTICLE 3 3.01	MANAGEMENT'S RIGHTS	1112121212
1.22 ARTICLE 2 2.01 ARTICLE 3 3.01 3.02 3.03	MANAGEMENT'S RIGHTS MANAGEMENT'S RIGHTS DISCRIMINATION NO DISCRIMINATION ACCESS TO PERSONNEL FILE SEXUAL HARASSMENT, PERSONAL HARASSMENT AND BULLYING	111212121212
1.22 ARTICLE 2 2.01 ARTICLE 3 3.01 3.02 3.03 ARTICLE 4	MANAGEMENT'S RIGHTS MANAGEMENT'S RIGHTS DISCRIMINATION NO DISCRIMINATION ACCESS TO PERSONNEL FILE SEXUAL HARASSMENT, PERSONAL HARASSMENT AND BULLYING UNION RECOGNITION	11121212121212
1.22 ARTICLE 2 2.01 ARTICLE 3 3.01 3.02 3.03 ARTICLE 4 4.01	MANAGEMENT'S RIGHTS MANAGEMENT'S RIGHTS DISCRIMINATION NO DISCRIMINATION ACCESS TO PERSONNEL FILE SEXUAL HARASSMENT, PERSONAL HARASSMENT AND BULLYING UNION RECOGNITION UNION RECOGNITION	
1.22 ARTICLE 2 2.01 ARTICLE 3 3.01 3.02 3.03 ARTICLE 4 4.01 4.02	MANAGEMENT'S RIGHTS MANAGEMENT'S RIGHTS DISCRIMINATION NO DISCRIMINATION ACCESS TO PERSONNEL FILE SEXUAL HARASSMENT, PERSONAL HARASSMENT AND BULLYING UNION RECOGNITION UNION ACTIVITIES	1112121212121313
1.22 ARTICLE 2 2.01 ARTICLE 3 3.01 3.02 3.03 ARTICLE 4 4.01 4.02 4.03	MANAGEMENT'S RIGHTS MANAGEMENT'S RIGHTS DISCRIMINATION NO DISCRIMINATION ACCESS TO PERSONNEL FILE SEXUAL HARASSMENT, PERSONAL HARASSMENT AND BULLYING UNION RECOGNITION UNION ACTIVITIES ACCESS TO CUPE NATIONAL REPRESENTATIVE	1112121212121313
1.22 ARTICLE 2 2.01 ARTICLE 3 3.01 3.02 3.03 ARTICLE 4 4.01 4.02 4.03 4.03 4.04	MANAGEMENT'S RIGHTS MANAGEMENT'S RIGHTS DISCRIMINATION NO DISCRIMINATION ACCESS TO PERSONNEL FILE SEXUAL HARASSMENT, PERSONAL HARASSMENT AND BULLYING UNION RECOGNITION UNION ACTIVITIES ACCESS TO CUPE NATIONAL REPRESENTATIVE NO OTHER AGREEMENTS	
1.22 ARTICLE 2 2.01 ARTICLE 3 3.01 3.02 3.03 ARTICLE 4 4.01 4.02 4.03 4.04 4.05	MANAGEMENT'S RIGHTS MANAGEMENT'S RIGHTS DISCRIMINATION NO DISCRIMINATION ACCESS TO PERSONNEL FILE SEXUAL HARASSMENT, PERSONAL HARASSMENT AND BULLYING UNION RECOGNITION UNION ACTIVITIES ACCESS TO CUPE NATIONAL REPRESENTATIVE NO OTHER AGREEMENTS BARGAINING UNIT WORK	111212121213131313
1.22 ARTICLE 2 2.01 ARTICLE 3 3.01 3.02 3.03 ARTICLE 4 4.01 4.02 4.03 4.04 4.05 4.06	MANAGEMENT'S RIGHTS MANAGEMENT'S RIGHTS DISCRIMINATION NO DISCRIMINATION ACCESS TO PERSONNEL FILE SEXUAL HARASSMENT, PERSONAL HARASSMENT AND BULLYING UNION RECOGNITION UNION ACTIVITIES ACCESS TO CUPE NATIONAL REPRESENTATIVE NO OTHER AGREEMENTS BARGAINING UNIT WORK JOB CREATION PROGRAMMES	111212121213131313
1.22 ARTICLE 2 2.01 ARTICLE 3 3.01 3.02 3.03 ARTICLE 4 4.01 4.02 4.03 4.04 4.05 4.06 4.07	MANAGEMENT'S RIGHTS MANAGEMENT'S RIGHTS DISCRIMINATION NO DISCRIMINATION ACCESS TO PERSONNEL FILE SEXUAL HARASSMENT, PERSONAL HARASSMENT AND BULLYING UNION RECOGNITION UNION ACTIVITIES ACCESS TO CUPE NATIONAL REPRESENTATIVE NO OTHER AGREEMENTS BARGAINING UNIT WORK JOB CREATION PROGRAMMES PICKET LINE	1112121212131313131414
1.22 ARTICLE 2 2.01 ARTICLE 3 3.01 3.02 3.03 ARTICLE 4 4.01 4.02 4.03 4.04 4.05 4.06 4.07 4.08	MANAGEMENT'S RIGHTS MANAGEMENT'S RIGHTS MISCRIMINATION NO DISCRIMINATION ACCESS TO PERSONNEL FILE SEXUAL HARASSMENT, PERSONAL HARASSMENT AND BULLYING UNION RECOGNITION UNION RECOGNITION UNION ACTIVITIES ACCESS TO CUPE NATIONAL REPRESENTATIVE NO OTHER AGREEMENTS BARGAINING UNIT WORK JOB CREATION PROGRAMMES PICKET LINE UNION TO NOTIFY CITY.	11121212131313131414
1.22 ARTICLE 2 2.01 ARTICLE 3 3.01 3.02 3.03 ARTICLE 4 4.01 4.02 4.03 4.04 4.05 4.06 4.07 4.08 ARTICLE 5	MANAGEMENT'S RIGHTS MANAGEMENT'S RIGHTS DISCRIMINATION NO DISCRIMINATION ACCESS TO PERSONNEL FILE SEXUAL HARASSMENT, PERSONAL HARASSMENT AND BULLYING UNION RECOGNITION UNION ACTIVITIES ACCESS TO CUPE NATIONAL REPRESENTATIVE NO OTHER AGREEMENTS BARGAINING UNIT WORK. JOB CREATION PROGRAMMES PICKET LINE UNION MEMBERSHIP	11121212131313141414
1.22 ARTICLE 2 2.01 ARTICLE 3 3.01 3.02 3.03 ARTICLE 4 4.01 4.02 4.03 4.04 4.05 4.06 4.07 4.08	MANAGEMENT'S RIGHTS MANAGEMENT'S RIGHTS MISCRIMINATION NO DISCRIMINATION ACCESS TO PERSONNEL FILE SEXUAL HARASSMENT, PERSONAL HARASSMENT AND BULLYING UNION RECOGNITION UNION RECOGNITION UNION ACTIVITIES ACCESS TO CUPE NATIONAL REPRESENTATIVE NO OTHER AGREEMENTS BARGAINING UNIT WORK JOB CREATION PROGRAMMES PICKET LINE UNION TO NOTIFY CITY.	1112121213131314141414

5.03 5.04	CITY TO ADVISE NEW EMPLOYEES	
5.05	Union Dues (T-4 Slips)	
ARTICLE 6	LABOUR/MANAGEMENT COMMITTEE	15
6.01	COMPOSITION OF COMMITTEE	
6.02	TERMS OF REFERENCE	
6.03	MEETINGS OF COMMITTEE	
6.04	MINUTES OF MEETINGS	
6.05	RECOMMENDATIONS OF COMMITTEE	16
ARTICLE 7	SENIORITY	16
7.01	Bargaining Unit Seniority	
7.02	SENIORITY DATE	
7.03	REGULAR PART-TIME EMPLOYEE	
7.04	RETENTION OF SENIORITY	
7.05	Loss of Seniority	
7.06	TRANSFER AND SENIORITY OUTSIDE BARGAINING UNIT	
7.07	SENIORITY LIST	
7.08	SENIORITY ACCUMULATION	
ARTICLE 8	PROBATION	
8.01	PROBATION	
8.02	NOTIFICATION OF STATUS	
8.03	Training Period	
ARTICLE 9	HOURS OF WORK	19
ARTICLE 9 9.01	HOURS OF WORK	
9.01 9.02	HOURS OF WORK	
9.01 9.02 9.03	HOURS OF WORK HOURS OF WORK MECHANIC'S SHIFTS ADVANCED HOURS	
9.01 9.02 9.03 9.04	HOURS OF WORK HOURS OF WORK MECHANIC'S SHIFTS ADVANCED HOURS NOTICE OF SHIFT CHANGE	
9.01 9.02 9.03 9.04 9.05	HOURS OF WORK HOURS OF WORK MECHANIC'S SHIFTS ADVANCED HOURS NOTICE OF SHIFT CHANGE PLACE TO COMMENCE SHIFTS	19 21 21 21
9.01 9.02 9.03 9.04 9.05 9.06	HOURS OF WORK. HOURS OF WORK. MECHANIC'S SHIFTS. ADVANCED HOURS. NOTICE OF SHIFT CHANGE. PLACE TO COMMENCE SHIFTS. REPORTING PAY GUARANTEE	19 21 21 21
9.01 9.02 9.03 9.04 9.05 9.06 9.07	HOURS OF WORK HOURS OF WORK MECHANIC'S SHIFTS ADVANCED HOURS NOTICE OF SHIFT CHANGE PLACE TO COMMENCE SHIFTS REPORTING PAY GUARANTEE REST PERIODS.	
9.01 9.02 9.03 9.04 9.05 9.06 9.07 9.08	HOURS OF WORK HOURS OF WORK MECHANIC'S SHIFTS ADVANCED HOURS NOTICE OF SHIFT CHANGE PLACE TO COMMENCE SHIFTS REPORTING PAY GUARANTEE REST PERIODS FLEXIBLE WORKING HOURS.	19 21 21 21 22 22
9.01 9.02 9.03 9.04 9.05 9.06 9.07 9.08 ARTICLE 10	HOURS OF WORK HOURS OF WORK MECHANIC'S SHIFTS ADVANCED HOURS NOTICE OF SHIFT CHANGE PLACE TO COMMENCE SHIFTS REPORTING PAY GUARANTEE REST PERIODS FLEXIBLE WORKING HOURS	192121212222
9.01 9.02 9.03 9.04 9.05 9.06 9.07 9.08 ARTICLE 10 10.01	HOURS OF WORK HOURS OF WORK MECHANIC'S SHIFTS ADVANCED HOURS NOTICE OF SHIFT CHANGE PLACE TO COMMENCE SHIFTS. REPORTING PAY GUARANTEE REST PERIODS. FLEXIBLE WORKING HOURS. OVERTIME OVERTIME CREDITS.	
9.01 9.02 9.03 9.04 9.05 9.06 9.07 9.08 ARTICLE 10 10.01 10.02	HOURS OF WORK HOURS OF WORK MECHANIC'S SHIFTS ADVANCED HOURS NOTICE OF SHIFT CHANGE PLACE TO COMMENCE SHIFTS REPORTING PAY GUARANTEE REST PERIODS FLEXIBLE WORKING HOURS OVERTIME OVERTIME CREDITS OVERTIME CALCULATION	
9.01 9.02 9.03 9.04 9.05 9.06 9.07 9.08 ARTICLE 10 10.01 10.02 10.03	HOURS OF WORK HOURS OF WORK MECHANIC'S SHIFTS ADVANCED HOURS NOTICE OF SHIFT CHANGE PLACE TO COMMENCE SHIFTS REPORTING PAY GUARANTEE REST PERIODS FLEXIBLE WORKING HOURS OVERTIME OVERTIME CREDITS OVERTIME CALCULATION OVERTIME ON A SUNDAY EQUIVALENT OR GENERAL HOLIDAY	
9.01 9.02 9.03 9.04 9.05 9.06 9.07 9.08 ARTICLE 10 10.01 10.02 10.03 10.04	HOURS OF WORK HOURS OF WORK MECHANIC'S SHIFTS ADVANCED HOURS NOTICE OF SHIFT CHANGE PLACE TO COMMENCE SHIFTS REPORTING PAY GUARANTEE REST PERIODS FLEXIBLE WORKING HOURS OVERTIME OVERTIME CREDITS OVERTIME CALCULATION OVERTIME ON A SUNDAY EQUIVALENT OR GENERAL HOLIDAY MEAL BREAKS	
9.01 9.02 9.03 9.04 9.05 9.06 9.07 9.08 ARTICLE 10 10.01 10.02 10.03 10.04 10.05	HOURS OF WORK HOURS OF WORK MECHANIC'S SHIFTS ADVANCED HOURS NOTICE OF SHIFT CHANGE PLACE TO COMMENCE SHIFTS REPORTING PAY GUARANTEE REST PERIODS FLEXIBLE WORKING HOURS OVERTIME OVERTIME CREDITS OVERTIME CALCULATION OVERTIME ON A SUNDAY EQUIVALENT OR GENERAL HOLIDAY MEAL BREAKS OVERTIME VOLUNTARY	
9.01 9.02 9.03 9.04 9.05 9.06 9.07 9.08 ARTICLE 10 10.01 10.02 10.03 10.04 10.05 10.06	HOURS OF WORK. HOURS OF WORK. MECHANIC'S SHIFTS. ADVANCED HOURS. NOTICE OF SHIFT CHANGE. PLACE TO COMMENCE SHIFTS. REPORTING PAY GUARANTEE. REST PERIODS. FLEXIBLE WORKING HOURS. OVERTIME. OVERTIME CREDITS. OVERTIME CALCULATION OVERTIME ON A SUNDAY EQUIVALENT OR GENERAL HOLIDAY. MEAL BREAKS. OVERTIME VOLUNTARY. BANKING OF OVERTIME	
9.01 9.02 9.03 9.04 9.05 9.06 9.07 9.08 ARTICLE 10 10.01 10.02 10.03 10.04 10.05	HOURS OF WORK HOURS OF WORK MECHANIC'S SHIFTS ADVANCED HOURS NOTICE OF SHIFT CHANGE PLACE TO COMMENCE SHIFTS REPORTING PAY GUARANTEE REST PERIODS FLEXIBLE WORKING HOURS OVERTIME OVERTIME CREDITS OVERTIME CALCULATION OVERTIME ON A SUNDAY EQUIVALENT OR GENERAL HOLIDAY MEAL BREAKS OVERTIME VOLUNTARY	

ARTICLE 12	ANNUAL VACATIONS	
12.01	Accrual Period	
12.02	VACATION ENTITLEMENT UPON TERMINATION OF EMPLOYMENT	
12.03	VACATION ENTITLEMENT	
12.04	VACATION PAY	
12.05	SCHEDULING VACATION	
12.06	MINIMUM ANNUAL VACATION	
12.07	Vacation Carry Over	
12.08	PAYMENT OF VACATION PAY	
12.09	ILLNESS OR DEATH DURING VACATION	
ARTICLE 13	GENERAL HOLIDAYS	
13.01	General Holidays	
13.02	SUBSTITUTED HOLIDAY	
13.03	SHIFT WORK	28
13.04	PAYMENT FOR WORK ON GENERAL HOLIDAYS	28
13.05	GENERAL HOLIDAY WHILE ON VACATION	28
13.06	Personal Day	28
13.07	TEMPORARY ASSIGNMENT	
ARTICLE 14	HEALTH AND WELFARE	. 29
14.01	Benefit Plans	29
14.02	PENSION (MUNICIPAL) PLAN	
14.03	SICK LEAVE	
14.04	Pregnancy Leave	
14.05	CONTINUATION OF BENEFITS	
14.06	LONG TERM DISABILITY	
14.07	LEAVE FOR BIRTH OR ADOPTION OF CHILD	
14.08	EMPLOYEE ASSISTANCE PROGRAM	36
14.09	ADOPTION LEAVE	36
14.10	Parental Leave	
14.11	SICK LEAVE AND ICBC OR WCB	
14.12	GENERAL LEAVE OF ABSENCE	
14.13	MEDICAL APPOINTMENTS	
14.14	FAMILY ILLNESS	
14.15	EMPLOYMENT INSURANCE REBATE	
14.16	FACILITY PASSES	38
ARTICLE 15	BEREAVEMENT LEAVE	. 39
15.01	BEREAVEMENT LEAVE	39
15.02	IMMEDIATE FAMILY	
15.03	Compensable Hours	
15.04	Pallbearer's and Mourner's Leave	39
15.05	COMPASSIONATE CARE LEAVE	39
ARTICLE 16	CONTRACTING OUT	. 39
16.01	RESTRICTIONS ON CONTRACTING OUT	40

16.02	ADVANCE NOTICE TO UNION	40
16.03	CONTRACTING OUT COMMITTEE	4C
16.04	JOINT CONTRACTING OUT COMMITTEE	40
ARTICLE 17	REMUNERATION	40
17.01	PAY SCHEDULES	40
17.02	Allowances	40
17.03	TEMPORARY ASSIGNMENT (HIGHER PAID POSITION)	42
17.04	TEMPORARY ASSIGNMENT (LOWER PAID POSITION)	42
17.05	JURY OR COURT WITNESS DUTY	
17.06	TOOL ALLOWANCE	42
17.07	BOOT ALLOWANCE	43
17.08	DAY OF OCCUPATIONAL INJURY	43
17.09	RECREATION DEPARTMENT	43
17.10	BATHING SUITS AND SHOES	43
17.11	RCMP Guards – Rest Periods	43
ARTICLE 18	PAYMENT SCHEDULE	
18.01	Pay Days	44
18.02	Pay Period	44
18.03	Pay on Termination	44
18.04	AUTOMATED PAYROLL DEPOSIT	44
ARTICLE 19	JOB RECLASSIFICATIONS	44
19.01	JOB DESCRIPTIONS	
19.02	New Positions	
19.03	CHANGES IN CLASSIFICATION	45
ARTICLE 20	GRIEVANCE PROCEDURE	45
20.01	GRIEVANCE DEFINED	
20.02	RIGHTS OF STEWARDS	
20.03	Rights of Grievor	
20.04	REPLIES IN WRITING	
20.05	DIFFERENCES TO BE DISCUSSED	
20.06	GRIEVANCE PROCEDURE	
20.07	GENERAL GRIEVANCE	47
20.08	CITY GRIEVANCE	47
20.09	GRIEVANCE ON SAFETY	47
20.10	Arbitration	47
20.11	AMENDING TIME LIMITS	48
20.12	Power to Amend	48
ARTICLE 21	TERMINATION AND DISCIPLINE	48
21.01	TERMINATION	48
21.02	DISCIPLINE	
21.03	REMOVAL OF REPRIMAND FROM PERSONNEL FILE	48
21.04	RIGHT TO ASSISTANCE FROM CUPE REPRESENTATIVES	49

ARTICLE 22	LAYOFF AND RECALL	49
22.01	LAYOFF GUIDELINES	. 49
22.02	Notice of Layoff	. 50
22.03	Bumping Procedure	.50
22.04	SEVERANCE PAY OPTIONS	
22.05	LAYOFF WHILE ON SICK LEAVE OR WCB	. 51
ARTICLE 23	EQUIPMENT CLASSIFICATION	52
23.01	LIGHT EQUIPMENT	
23.02	HEAVY EQUIPMENT	. 52
23.03	HIAB TRUCK	. 52
23.04	SPECIAL EQUIPMENT	. 52
23.05	New Equipment	. 52
ARTICLE 24	PROTECTIVE CLOTHING	52
24.01	THE CITY SHALL PROVIDE PROTECTIVE CLOTHING AND SAFETY EQUIPMENT IN	
	Following Manner:	
24.02	ALL OTHER ITEMS OF CLOTHING	. 53
ARTICLE 25	VACANCIES	
25.01	Posting of Vacancies	
25.02	TEMPORARY POSTINGS	
25.03	Acting Foreman	
25.04	TEMPORARY ASSIGNMENTS	. 54
ARTICLE 26	TECHNOLOGICAL CHANGE	
26.01	ADVANCE NOTICE	
26.02	Changes	
26.03	NO AGREEMENT ON RESULTS OF THE CHANGE	
26.04	REGULAR EMPLOYEE	
26.05	TRAINING PERIOD FOR TECHNOLOGICAL CHANGES	
26.06	DISPLACED EMPLOYEES	
ARTICLE 27	UNION DUTY	
27.01	Union Leaves With Pay	
27.02	NO OVERTIME ON UNION LEAVES WITH PAY	
27.03	UNION LEAVES WITHOUT PAY	. 55
ARTICLE 28		
MIZIAWEE EU	GENERAL CONDITIONS	56
28.01	CHANGES TO AGREEMENT	. 56
	CHANGES TO AGREEMENT	. 56 . 56
28.01 28.02 28.03	CHANGES TO AGREEMENT PRINTING OF AGREEMENT LOSS OF EMPLOYEE QUALIFICATIONS	. 56 . 56 . 57
28.01 28.02 28.03 28.04	CHANGES TO AGREEMENT PRINTING OF AGREEMENT LOSS OF EMPLOYEE QUALIFICATIONS QUALIFICATIONS	. 56 . 56 . 57 . 57
28.01 28.02 28.03 28.04 28.05	CHANGES TO AGREEMENT PRINTING OF AGREEMENT LOSS OF EMPLOYEE QUALIFICATIONS QUALIFICATIONS WORKERS' COMPENSATION PENSION.	. 56 . 56 . 57 . 57
28.01 28.02 28.03 28.04 28.05 28.06	CHANGES TO AGREEMENT PRINTING OF AGREEMENT LOSS OF EMPLOYEE QUALIFICATIONS QUALIFICATIONS WORKERS' COMPENSATION PENSION. DAY OF MOURNING	. 56 . 57 . 57 . 57 . 57
28.01 28.02 28.03 28.04 28.05 28.06 28.07	CHANGES TO AGREEMENT PRINTING OF AGREEMENT LOSS OF EMPLOYEE QUALIFICATIONS QUALIFICATIONS WORKERS' COMPENSATION PENSION. DAY OF MOURNING IN-SERVICE TRAINING (AQUATIC STAFF)	. 56 . 57 . 57 . 57 . 57 . 58
28.01 28.02 28.03 28.04 28.05 28.06	CHANGES TO AGREEMENT PRINTING OF AGREEMENT LOSS OF EMPLOYEE QUALIFICATIONS QUALIFICATIONS WORKERS' COMPENSATION PENSION. DAY OF MOURNING	. 56 . 57 . 57 . 57 . 57 . 58

ARTICLE 29	TERM OF AGREEMENT	58
29.01	TERM OF AGREEMENT	58
SCHEDULE "A"	′ – WAGES	59
	′ – WAGES	
LETTER OF UN	IDERSTANDING NO. 1	62
RE: MAN	AGEMENT STAFF ON LTD PLAN	63
LETTER OF UN	IDERSTANDING NO. 2	63
Re: Char	NGE OF SHIFT PUBLIC WORKS DEPARTMENT - SCHEDULE A EMPLOYEES	64
	IDERSTANDING NO. 3	
Re: Arti	CLE 17.02 (D)	65
LETTER OF UN	DERSTANDING NO. 4	65
RE: WAGE	REVIEW FOR RED SEAL TICKETED STAFE AND ARTICLE 17 02 ALLOWANCES	66

AGREEMENT BETWEEN

THE CITY OF TERRACE

(hereinafter called the "City")

and

THE CANADIAN UNION OF PUBLIC EMPLOYEES LOCAL 2012

(hereinafter called the "Union")

The object to this Agreement is to:

- (1) Promote continuing harmonious relations, co-operation and understanding between the City and its employees, with a view to exercising the terms of this Agreement in a fair and reasonable manner.
- (2) Provide for conditions of employment, hours of work and rates of pay.
- (3) Provide for the prompt, fair and peaceful settlement of disputes or grievances.
- (4) Encourage efficiency in operations.

ARTICLE 1 DEFINITIONS

1.01 Approved Absence

Approved absence shall be: Any absence authorized under the terms of this contract and shall include service with Armed Forces during a national emergency; leave to run for or serve in any Public or Union Office, provided that thirty (30) days' notice shall be given prior to the employee's return to work after such service.

1.02 Callout

"Callout" is that occasion when an employee is required by the City to return to work for the City outside of the hours of their normal working shift in order to perform work for the City.

1.03 Chargehand

"Chargehand" shall mean an employee who is designated by a Supervisor to supervise the work of employee(s) in an assigned group of one (1) or more co-employees, for the duration of a particular assignment.

1.04 Continuous Service

"Continuous Service" shall mean a period of employment with the City which is uninterrupted, except by an approved absence.

1.05 Employee

"Employee" shall have the same meaning as defined in the Labour Relations Code of British Columbia.

1.06 Regular Full-time Employee

"Regular Full-time Employee" shall mean an employee who has successfully completed a probation period of sixty-five (65) days of work, within any consecutive six (6) month period, in a position covered by this Agreement and is employed full-time.

1.07 Regular Part-time Employee

"Regular Part-time Employee" shall mean an employee who has successfully completed a probation period of sixty-five (65) days of work, within any consecutive six (6) month period, in a position covered by this Agreement and is employed sixty (60) or more hours over a four (4) week period.

Any employee who has worked full time hours for a six (6) month period will be reclassified as a regular full-time employee.

The Employer shall provide the Union notice, in writing, when the change of status is applied.

1.08 Auxiliary Lifeguard/Instructor

Auxiliary Lifeguard Instructor shall mean an Aquatic Centre employee whose hours of work shall be regularly scheduled, and who works less than twenty (20) hours every week. These employees shall not be eligible for benefits under **Article 14**.

1.09 Temporary and Seasonal Employees

"Temporary Employee" shall mean an employee appointed to a position with a specified end date or appointed to a position with an unspecified end date due to sick leave replacement. This may include full-time and part-time employees.

Temporary Employees in a temporary position of more than three (3) months will accrue seniority from the date of hire upon passing of probation and will be entitled to all benefits of this collective agreement.

"Seasonal Employee" shall mean an employee whose anticipated length of employment shall not exceed seven (7) calendar months. This may include full-time and part-time employees. Seasonal employees shall be limited to Seasonal Parks or Seasonal Construction positions only.

1.10 Casual Employee

"Casual Employee" shall mean an employee hired on a day-to-day basis who works less than sixty (60) hours every four (4) weeks.

1.11 Student Employee

A Student Employee shall mean a student hired on a part-time or full-time basis between May 1st and the Friday immediately before Labour Day. A student is defined as a person who intends to continue their education on a full-time basis in September. The City shall advise the student, at the time of appointment, of the anticipated date of termination. Notwithstanding Article 22, notice of layoff is not required.

A Student Employee shall receive the benefits accorded a temporary employee.

When the appointee is already on staff they shall continue to enjoy those benefits to which they were entitled with the exception of seniority accumulation and retention and that there is no requirement for notice of layoff. Student employees shall not be appointed or retained when regular employees on layoff are capable (qualified) to perform the work.

1.12 Layoff

A "Layoff" is defined as the reduction of a regular employee's regular hours of work or a reduction in the work force, but does not include the splitting of a job by mutual agreement between the City and the Union, so as to create two jobs.

1.13 Overtime

"Overtime" shall mean any hours worked in excess of those defined in Article

1.14 Overtime Credits

Each overtime credit shall equal one (1) hour's time or value therefore at the employee's regular base rate.

1.15 Personal Harassment

Personal harassment shall be defined as verbal threats and/or verbal abuse, derogatory comments that ought reasonably to be known to be offensive, physical threats and/or physical abuse, and/or intimidation or actions that demean and belittle an individual or cause personal humiliation.

1.16 Probation

"Probation" shall mean the period of time between the date of commencement of employment and the date on which seniority is granted.

1.17 Definitions - Seniority

"Seniority" shall mean the accumulated lengths of time specified in Articles 7.02, 7.08, 14.04, 14.11, and 15.05 and does not include service as an excluded employee.

1.18 Sexual Harassment

"Sexual Harassment" shall be defined as any comment or conduct of a sexual nature that is known or ought to be reasonably known to be unwelcome and shall include, but is not limited to:

- (a) sexual solicitation or advances; inappropriate touching or sexual comments; or
- (b) any threat or reprisal which might reasonably be perceived as placing a condition on employment by a person in authority after the above conduct has been rejected.

1.19 Sunday Equivalent

"Sunday equivalent" shall mean the second (2nd) day of a regularly scheduled two (2) day rest period and the second (2nd) and fourth (4th) days of a regularly scheduled four (4) day rest period of an employee whose work week differs from a normal calendar week.

1.20 Technological Change

- (a) the introduction by the City of a change in its work, undertaking or business, or a change in its equipment or material from the equipment or material previously used by the City in its work, undertaking or business; or
- (b) a change in the manner the City carries on its work, undertaking or business related to the introduction of that equipment or material.

1.21 Week

"Week" shall mean a period between "midnight" on Friday night and "midnight" on the succeeding Friday night.

1.22 Spouse

A person to whom the employee is married or is living with another person in a marriage-like relationship, as defined in the British Columbia Family Law Act.

This definition shall determine all other familial relationships referred to in this agreement, including, but not restricted to, "child," which shall include the employee's partner's child, and the definition of "in-law," which shall include equivalent relationships flowing from **marriage or marriage-like** relationships.

ARTICLE 2 MANAGEMENT'S RIGHTS

2.01 Management's Rights

The Union recognizes the right of the City to manage the affairs of its business and direct its working forces subject to the terms of this Agreement. Such right includes but is not limited to the following:

(1) hire (2) assign work (3) grant regular status (4) suspend or discharge for just cause (5) promote (6) demote (7) discipline for just cause (8) lay off (9) transfer (10) reinstate (11) assess an employee's skills, competence, efficiency and qualifications (12) develop and administer tests to assist in the adjudication of an employee's skills, competence, efficiency and qualifications.

The Parties agree that the addition of (12) does not confer any new rights to the City and does not diminish the right of the Union to grieve testing.

ARTICLE 3 DISCRIMINATION

3.01 No Discrimination

The City and the Union agree that there shall be no discrimination or coercion exercised or practised with respect to any employee in the matter of hiring, wage rates, training, upgrading, promotion, transfer, layoff, recall, discipline, classification, discharge or otherwise by reason of age, race, creed, colour, national origin, religion, political affiliation or activity, sexual orientation, sex or marital status, employee place of residence, nor by reason of their membership or activity in the Union.

3.02 Access to Personnel File

An employee shall have access to their personnel file. Review of personnel files shall be scheduled with and in the presence of the Chief Administrative Officer or **their** designate. An employee shall receive copies of all documents placed in their personnel file on behalf of the City.

3.03 Sexual Harassment, Personal Harassment and Bullying

The Parties recognize the right of CUPE members to work in an environment free from sexual harassment and agree to cooperate in attempting to resolve, in a confidential manner, any complaints of sexual harassment which may arise in the workplace.

The Parties also recognize the right of CUPE members to work in an environment free from personal harassment or bullying.

The Employer is committed to taking appropriate disciplinary action if an allegation of sexual harassment, personal harassment and/or bullying is substantiated.

Cases of sexual harassment, personal harassment, and bullying shall be considered as discrimination and are eligible to be processed as a grievance and/or through the Human Rights Tribunal.

In cases where sexual harassment, personal harassment or bullying may result in the transfer of an employee, the victim shall not be transferred against their will.

When an allegation of sexual harassment, personal harassment and/or bullying is proven to be malicious or frivolous in nature, the complainant may be subject to appropriate discipline.

ARTICLE 4 UNION RECOGNITION

4.01 Union Recognition

- (a) The City recognizes the Canadian Union of Public Employees, Local 2012, as the collective bargaining agent for all employees of the City for whom the Union has been certified as bargaining agent under the Labour Relations Code of British Columbia.
- (b) As soon as reasonably possible, the City shall notify, in writing, the Union whenever there is a change in employment status of its member(s) This includes whenever an employee is terminated, as well as a promotion or demotion.

4.02 Union Activities

The administration of Union duties, by an employee, such as typing, photocopying, mailing and attending meetings shall not be carried out during working hours, except as provided in Article 27.

4.03 Access to CUPE National Representative

The Union shall have the right at any time to have the assistance of representatives of the Canadian Union of Public Employees or any other advisors when dealing or negotiating with the City. Such representative(s)/advisor(s) shall have access to the employer's premises in order to deal with matters arising out of this Collective Agreement, provided permission has been granted. Such permission shall not be unreasonably denied.

4.04 No Other Agreements

No employee shall be required or permitted to make a written or verbal agreement with the City which may conflict with the terms of this Agreement.

4.05 Bargaining Unit Work

Employees of the City whose jobs are not in the bargaining unit shall not work on any jobs which are included in the bargaining unit, except in cases mutually agreed upon by the City and the Union and in emergency and training programmes.

4.06 Job Creation Programmes

The City, by special written agreement with the Union, may enter into a contract with either the Provincial or Federal government to participate in job creation programmes. No project shall commence until said contract is ratified by both parties. Said written agreement with the Union shall deal with:

- (1) Job tasks
- (2) Rates of pay
- (3) Hours of work
- (4) Employee benefits
- (5) Equipment and transportation
- (6) Supervision
- (7) Union membership
- (8) Seniority

4.07 Picket Line

Pursuant to the Labour Relations Code of British Columbia, an employee covered by this Agreement shall have the right to refuse to cross a picket line arising out of labour disputes.

4.08 Union to Notify City

As soon as reasonably possible, after the election of Executive Officers, Shop Stewards and committee members, the Union shall notify the City in writing of the persons filling the positions.

ARTICLE 5 UNION MEMBERSHIP

5.01 Union Membership

All employees covered by this Agreement shall become and remain a member of the Union as a condition of employment within fifteen (15) calendar days of the date of hiring.

5.02 Union Dues Deduction

Starting with the second pay period after the date of hire, the City shall commence deducting from the wages or salary of the employee, initiation fees, dues or assessments levied in the amount specified by the Union, and shall forward such monies to the Secretary-Treasurer of the Union before the

fifteenth (15th) day of the following month, accompanied by a list of employees from whom the deductions were made.

5.03 City to Advise New Employees

The City agrees to advise new employees of the existence of this Union Agreement, the conditions of employment, safety rules and all benefit plans.

5.04 City to Introduce New Employees to Union Steward

The new employee's immediate supervisor shall introduce the employee to **their Union representative**, as soon as possible after **they** commence employment.

The City shall allow up to one (1) hour per employee, without loss of seniority, pay or benefits, for a Union representative to orient a newly hired person into the bargaining unit.

5.05 Union Dues (T-4 Slips)

The City shall cause to be shown on each employee's income tax T-4 slip the amount of the Union dues paid by the employee for the calendar year for which the T-4 slip is issued.

ARTICLE 6 LABOUR/MANAGEMENT COMMITTEE

6.01 Composition of Committee

The Labour/Management Committee shall consist of three (3) members of Management, three (3) members of CUPE Local 2012, and shall function within the limitations of the terms of reference mutually agreed between the City and the Union. When requested, CUPE Local 2012 shall have the assistance of the CUPE National Servicing Representative at such meetings, who shall have voice but no vote at the meeting. The Employer may also have the ability to have one advisor with voice, but no vote, at the Committee meetings. Labour Management Committee meetings shall not be adjusted solely to accommodate the attendance of such advisors.

6.02 Terms of Reference

The membership of the Labour/Management Committee shall collectively set its own terms of reference subject to ratification by the City and by the membership of the Union.

6.03 Meetings of Committee

The Committee shall meet no less than once each month on a regular basis.

6.04 Minutes of Meetings

The minutes of the Labour/Management Committee meetings shall be signed by a representative of each participating party and shall be posted on all employee bulletin boards.

6.05 Recommendations of Committee

The Labour/Management Committee findings may be recommended to the Union and the City for ratification.

ARTICLE 7 SENIORITY

7.01 Bargaining Unit Seniority

Seniority shall operate on a bargaining unit-wide basis.

Seniority shall apply in cases of promotion, demotion, transfer, layoff, bumping and rehiring, as the deciding factor when the skills, competence, efficiency and qualifications of the employees concerned are equal.

7.02 Seniority Date

An employee shall not achieve seniority until they have completed a probationary period at which time seniority shall date back to the date of hiring.

For purposes of determining seniority accumulation, an employee who was previously employed shall have those hours worked added to their accumulation, provided the periods of employment are consistent with Articles 7.04 and 7.05. Upon satisfactory completion of a period of employment as a casual and/or temporary employee, such an employee shall be considered laid off unless the employee has self-terminated.

7.03 Regular Part-time Employee

A regular part-time and casual recreation and casual RCMP guard employees shall accumulate seniority on the basis of actual hours worked converted to equivalent working days, based on the regular full-time hours for that classification.

7.04 Retention of Seniority

- (a) An employee who is laid off after less than one (1) year of service shall retain their earned seniority for a period of six (6) months, or the length of their service, whichever is greater.
- (b) An employee who is laid off after one (1) year of service shall retain seniority for a period of one (1) year.
- (c) Seniority shall not accumulate during a period of layoff.

7.05 Loss of Seniority

An employee shall lose their seniority for any of the following reasons:

- (a) Upon voluntarily leaving the service of the City;
- (b) When discharged for just cause;
- (c) When continuously laid off for a period exceeding the time factors as provided in Article 7.04.
- (d) An employee shall forfeit their seniority on the casual seniority list if they voluntarily leave a casual position for a regular position.
- (e) An employee shall forfeit their seniority on the regular seniority list if they voluntarily leave a regular position for a casual position.

7.06 Transfer and Seniority Outside Bargaining Unit

No employee shall be transferred to a position outside the bargaining unit without their consent. If an employee is transferred to a position outside of the bargaining unit, they shall retain their seniority accumulated up to the date of leaving the unit. Such employee shall have the right to return to their former position in the bargaining unit within six (6) months of leaving the unit and such period shall be at no loss in seniority or benefits.

Any other employee promoted or transferred because of the rearrangement of positions shall also be returned to their former position, or to another position with no loss in pay and which is mutually agreeable between the City and the employee.

7.07 Seniority List

- (a) The City shall maintain regular employee and casual recreation and casual guard employee seniority lists showing the date upon which each employee's service commenced, and the accumulated seniority since that date, with each list sorted in the order of accumulated seniority. There shall be one (1) seniority list for regular employees and one (1) for casual employees in the recreation department, and one (1) for casual guards.
- (b) The employees on the casual guard seniority list shall be called in on a rotational basis, in order to divide the work equitably between those employees. The employees on the casual recreation seniority list shall be called in and scheduled on as equitable basis as possible based on their availability.
 - Regular employees (not laid off employees) who want to take casual (not temporary) work instead of their regularly scheduled work shall be considered to be a casual with casual seniority and lose their regular seniority.
- (c) Seniority for casual recreation employees and casual guards shall be backdated to date of hire upon completion of their probationary period.

- (d) An employee on the casual seniority list shall have first consideration over external applicants for temporary, casual and regular positions in the same kind of work, but shall not have preference over one on the regular employee seniority list.
- (e) An up-to-date seniority list shall be sent to the Union and posted on all bulletin boards in April and October of each year.

7.08 Seniority Accumulation

Seniority shall continue to accumulate while an employee is:

- (a) in receipt of wages, on vacation or paid leave;
- (b) absent due to illness or accident;
- (c) absent for up to five (5) consecutive days unpaid personal leave;
- (d) on Union duty leave
- (e) on pregnancy leave as provided in Article 14.04 (a);
- (f) on parental leave as provided in Article 14.11;
- (g) on compassionate care leave as provided in Article 15.05;
- (h) while off on Worker's Compensation.

In all other circumstances, accumulated seniority shall be retained and shall be subject to Article 7.05.

ARTICLE 8 PROBATION

8.01 Probation

A new employee shall be on probation for sixty-five (65) days of work, within any consecutive six (6) month period, from the date of hiring. By mutual agreement between the City and the Union, an employee's probation period may be extended for up to thirty (30) additional days of work and, if necessary, the six (6) month period shall be waived.

8.02 Notification of Status

Upon completion of a probation period, the City shall notify the person in writing of termination or granting of seniority, as the case may be.

8.03 Training Period

An employee who is promoted, demoted (employee requested), or transferred to another position shall be a trainee with respect to the new position for a period of sixty-five (65) working days. Failure to satisfactorily complete the training period shall result in reinstatement to their former position.

By mutual agreement between the City and the Union, an employee's training period may be extended.

Any other employee promoted or transferred because of the rearrangement of positions shall also be returned to their former position, or to another position with no loss in pay and which is mutually agreeable between the City and the employee.

At any time during the training period, an employee may request, in writing, to be returned to **their** former position. The Union, the employee and the City shall meet to consider the request, and if the employee's concerns cannot be overcome by mutual agreement, the City shall concur with the request.

ARTICLE 9 HOURS OF WORK

9.01 Hours of Work

The regular work week and work day shall be as follows:

(a) <u>Schedule A Employees (Normal Operation)</u>

Eight (8) hours per day between the hours of 8:00 a.m. and 4:30 p.m., exclusive of a one-half ($\frac{1}{2}$) hour meal period, Monday to Friday inclusive.

(b) Schedule A Employees (Shift Work)

Eight (8) consecutive hours within a twenty-four (24) hour period, including a one-half ($\frac{1}{2}$) hour meal period for forty (40) working hours per week between Saturday and Friday. There shall be a minimum of ten (10) hours between shifts.

(c) Schedule B Employees (Normal Operation)

Seven (7) hours per day between the hours of 8:00 a.m. and 5:00 p.m., exclusive of a one (1) hour meal period, Monday to Friday inclusive.

(d) RCMP Secretary

Seven (7) consecutive hours per day between the hours of 8:00 a.m. and 5:00 p.m., exclusive of a one (1) hour meal period, for five (5) consecutive days from Monday to Saturday inclusive.

(e) RCMP – Records Coordinator

Seven (7) consecutive hours per day between the hours of 7:00 a.m. and 3:00 p.m., exclusive of a one (1) hour meal period, from Monday to Friday inclusive.

(f) Shifts

Day Shift shall mean eight (8) working hours commencing at 8:00 a.m. daily.

Afternoon Shift shall mean eight (8) working hours commencing at 4:00 p.m. daily.

Animal Shelter Afternoon Shift shall mean eight (8) working hours commencing at 11:00 a.m.

Night Shift shall mean eight (8) working hours commencing at 12:00 midnight daily.

Early Morning Shift shall mean eight (8) working hours commencing at 5:00 a.m. for street sweeping and winter road maintenance and asphalt patching.

(g) RCMP Guards

The normal work shift shall consist of twelve (12) hours, inclusive of meal breaks, (which are to be taken at the worksite), for four (4) consecutive days, followed by four (4) days of rest so as to average forty-two (42) working hours per week. An employee shall work two (2) day shifts followed by two (2) night shifts. Shift premiums apply as appropriate.

(h) Recreation Attendants

When the ice is in the Arena, persons employed for ice maintenance may be scheduled to work ten (10) hours, inclusive of meal breaks (which are to be taken at the worksite), as follows:

6:00 a.m. to 4:00 p.m. or 4:00 p.m. to 2:00 a.m.

An employee shall work four (4) such consecutive shifts followed by three (3) days of rest so as to average forty (40) hours of work per week. Shift premiums apply as appropriate.

(i) Aquatic Centre Employees

Regular part-time and full-time employees working at the Aquatic Centre may have their shifts commence at times other than specified in Article 9.01 (f). Aquatic Centre employees working such shifts shall be paid appropriate shift premiums.

Employees at the Aquatic Centre shall be given one (1) week's advance notice of new shift schedules. Casual employees shall give notice of their availability three (3) weeks in advance of the new shift schedule.

(j) Schedule A and B Employees

By mutual agreement between an employee and their Department Head, an employee may have their regular starting hours of work scheduled for:

7:00 a.m. or 7:30 a.m. or 8:00 a.m. or 8:30 a.m. or 9:00 a.m.

This agreement shall be in writing and shall not be unreasonably denied.

(k) Animal Shelter Employees

Eight (8) hours per day between the hours of 8:00 a.m. and 5:00 p.m. exclusive of one-half ($\frac{1}{2}$) hour meal period, for five (5) consecutive days, Monday to Sunday inclusive.

(I) Maximized Hours

Regular employees working less than full-time hours will have their hours maximized to full-time whenever there are extra hours in the same classification. Hours will not be added to casual or temporary employees or used to hire new employees until regular employees have had their hours maximized to the maximum of their posted hours.

9.02 Mechanic's Shifts

By mutual agreement between the City and a Mechanic, a shift may commence at 10:00 a.m. and end at 6:30 p.m. By mutual agreement between the City and the Union, other hours may be worked. Shift premiums shall apply as appropriate.

9.03 Advanced Hours

By mutual agreement, the hours in Article 9.01 may be advanced up to two (2) hours.

9.04 Notice of Shift Change

- (a) The City shall have the right to change an employee's shift, provided that the employee is given twenty-four (24) hours' notice of such change and that notice is verified as received by the employee affected and provided there is a minimum break of twelve (12) hours between the end of one shift and the start of the next. Failure to provide the required notice and/or break shall result in overtime rates of pay for the subsequent shift. By mutual agreement between an employee and their supervisor other arrangements for giving notice may be made (i.e., voice mail message).
- (b) The City shall have the right to change a casual employee's shift and shall give the employee twenty-four (24) hours' notice. An employee requesting a shift change shall give the employer twenty-eight (28) hours' notice of the requested change.

9.05 Place to Commence Shifts

Shifts shall commence and end at the appropriate City facility.

9.06 Reporting Pay Guarantee

A regular employee reporting for work on their regular shift shall be paid their regular rate of pay for all hours worked with a minimum of two (2) hours' pay if they do not commence work and a minimum of four (4) hours' pay if they commence work.

Casual employees called in and reporting for work shall be paid a minimum of three (3) hours' pay if they commence work and two (2) hours' pay if they do not commence work, with the exception of a casual high school or college student called in and reporting for work shall be paid a minimum of two (2) hours' pay.

9.07 Rest Periods

Fifteen (15) minute rest periods shall be taken on the job site during each shift. The timing of the rest periods shall be determined by the job supervisor. The first to be approximately two (2) hours after the start of the shift and subsequent breaks shall be approximately every two (2) hours thereafter unless the break falls on a lunch break.

9.08 Flexible Working Hours

- (a) The Aquatic Foreman, the Programs Supervisor, and the Leisure Services Programmer shall be permitted flexible working hours, as may be mutually agreed upon by the employee and Department Head. Overtime credits shall not apply for pre-arranged activities with the exception of work on a general holiday. In all other situations all terms of the collective agreement shall apply.
- (b) Employees may request, with the approval of the Department Head, on an exceptional basis, to adjust their work hours to accommodate personal obligations. Such requests will be based on the operational needs of the department and will not bear any extra costs to the City.

ARTICLE 10 OVERTIME

10.01 Overtime Credits

Overtime credits shall be accumulated for all hours worked in excess of those defined in Article 9.01. Overtime credits shall accumulate in one-half ($\frac{1}{2}$) hour increments.

10.02 Overtime Calculation

(a) Overtime hours worked during any one twenty-four (24) hour period commencing with the regular shift, or overtime hours worked in excess of the hours specified in Article 9 for the regular work week shall accumulate overtime credits at one and one-half (1½) times the employee's basic hourly rate for the first three (3) hours and double the employee's basic hourly rate thereafter. Employees required to work

overtime beyond their regular shift shall be given a paid fifteen (15) minute rest period at the end of the overtime shift. When the length of the overtime is anticipated to be more than two (2) hours a paid fifteen (15) minute rest period may be taken prior to the commencement of the overtime.

(b) An employee shall not be eligible for overtime credits if, by mutual agreement between the employee and their supervisor, the employee works beyond the normal end of shift to complete their full regular hours for that day.

10.03 Overtime on a Sunday Equivalent or General Holiday

Overtime hours worked on a Sunday equivalent as defined in Article 1.20 or a General Holiday shall accumulate overtime credits at double the employee's basic hourly rate. All overtime hours worked in an overtime shift commencing on a Sunday equivalent or General Holiday shall accumulate credits at double the employee's basic hourly rate.

10.04 Meal Breaks

- (a) Each employee shall be permitted a one-half (½) hour break for meal consumption for every five (5) hours worked. To qualify for the second (2nd) and subsequent meal breaks, an employee must work in excess of each five (5) hours.
- (b) For the second (2nd) and subsequent meal breaks, the time shall be paid, and a **twenty dollar (\$20.00)** meal ticket shall be provided at City expense for those employees not on a regularly scheduled ten (10) or twelve (12) hour shifts. Payment in lieu of taking a meal is subject to approval of the Department Head.

10.05 Overtime Voluntary

Overtime, standby and call-back time shall be divided equitably among employees who are willing and qualified to perform the available work. Overtime and standby shall be on a voluntary basis.

10.06 Banking of Overtime

Overtime credits shall, at the discretion of the employee, be paid in cash or, subject to Article 12.05, be taken as time off with pay, as one (1) hour's pay for each overtime credit accumulated. Overtime credits, up to **one hundred sixty (160)** hours may be carried over for use in the next calendar year. Overtime credits over **one hundred sixty (160)** hours may be accumulated for future use subject to approval by the Department Head. Overtime credits may be **taken or** paid out at any time at the discretion of the employee.

ARTICLE 11 EMERGENCY SERVICES

11.01 Standby Provision

- (a) An employee who is required to be available on call (standby) for any emergency which may arise shall receive two (2) overtime credits for each eight (8) hours of standby time, with a minimum of an eight (8) hour block of standby time.
- (b) An employee who is required to work during a standby period shall receive overtime credits as per Article 10 for all hours worked. "Call out" time shall not be less than two (2) hours. The employee shall not accumulate further overtime credits during the paid minimum of two (2) hours.

11.02 Callout

An employee called out who is not on standby, shall accumulate overtime credits for all hours worked, with a minimum of three (3) hours and shall not accumulate further overtime credits during the paid minimum of three (3) hours.

ARTICLE 12 ANNUAL VACATIONS

12.01 Accrual Period

For the purpose of Article 12, vacation entitlement shall be based on the **twelve (12) month period running from January 1**st **to December 31**st. An employee not actively at work but still continuing to accumulate seniority under Article 7.08 will be considered actively at work for the purposes of calculating vacation entitlement under this Article. Time worked in a calendar year January 1st to December 31st shall count as a year worked for vacation entitlement in Article 12.03.

For the purpose of calculating the number of years of service for determining vacation entitlement under Article 12.03, for employees hired prior to September 1st, vacation year number one (1) will be the year of hire. For employees hired between September 1st and December 31st, vacation year number one (1) will be calculated from January 1st following the date of hire.

Employees hired prior to January 1st, 2000 shall have their year of hire considered vacation year number one (1).

12.02 Vacation Entitlement Upon Termination of Employment

a) An employee who self-terminates their service with the City or is discharged for just cause before they have completed one full year of service from the date of hire to termination, shall be granted four percent (4%) vacation pay only. If vacation time taken exceeds the four percent (4%) entitlement, the amount of such unearned advance will be repaid by the employee and

- the City will have the authority to collect such overpayment by payroll deduction.
- b) Upon termination of employment, an employee shall receive vacation for the calendar year in which termination occurs on the basis of their monthly entitlement for that year, defined in Article 12.03 for each month or partial month worked to the date of termination.
- c) If an employee's employment terminates after the advanced vacation is taken but before such vacation has been earned, the amount of such unearned advance will be repaid by the employee and the City will have the authority to collect such overpayment by payroll deduction.

12.03 Vacation Entitlement

Vacation days are credited to <u>regular</u> employees prior to being earned at the beginning of the year.

Pro-rated vacation days for the first partial year of service will be credited to employees upon successful completion of the probationary period. An employee on probation shall not be allowed to take vacation leave with pay during the probationary period.

Entitlement shall be:

- (a) In the first (1st) year of service **up to and including the fifth (5th) year of service,** vacation entitlement shall be fifteen (15) working days.
- (b) In the sixth (6th) vacation year and up to and including the **twelfth** (12th) vacation year of continuous service, twenty (20) working days shall be granted.
- (c) In the **thirteenth (13th)** vacation year up to and including the nineteenth (19th) vacation year of continuous service, twenty-five (25) working days shall be granted.
- (d) In the twentieth (20th) vacation year and up to the twenty-fourth (24th) vacation year of continuous service, thirty (30) working days shall be granted.
- (e) In the twenty-fifth (25th) vacation year and thereafter, one (1) additional working day per additional vacation year of service.
- (f) Retiring employees may use vacation earned up to the date of retirement prior to their retirement date, if requested at minimum one (1) month to the anticipated date of vacation.

12.04 Vacation Pay

Vacation pay shall be paid at the employee's regular rate of pay, at the time the vacation is taken, or at a percentage of **their** gross pay for the previous whole or part calendar year, whichever is greater.

Vacation entitlement up to and including fifteen (15) days – Six percent (6%) Vacation entitlement of twenty (20) days - Eight percent (8%) Vacation entitlement of twenty-five (25) days - Ten percent (10%) Vacation entitlement of thirty (30) days - Twelve percent (12%)

Plus an additional zero point four percent (0.4%) for each additional day of vacation entitlement.

12.05 Scheduling Vacation

Vacation requests are to be submitted by November 30th for the next calendar year to the appropriate Department Head, or their designate. Seniority shall be the basis on which the decision shall be reached for the first (1st) priority choice, if the vacation period of two (2) or more employees within the same department conflicts. All requests will be approved or denied by December 31st.

Any vacation requests that are submitted after this date will be looked at on a first-come-first-served basis and based on the operational needs of the department. These vacation requests shall be responded to within two (2) weeks of the request and shall not be unreasonably denied. Both approval and denial of vacation shall be in writing, with those being denied accompanied by a reasonable explanation.

Annual vacation entitlements must be scheduled prior to an unpaid leave of absence being approved.

12.06 Minimum Annual Vacation

An employee may receive vacation pay in lieu of vacation time when through extenuating circumstances they are unable to take their vacation.

Pay in lieu of vacation shall not be granted for the first ten (10) working days of entitlement and for the first fifteen (15) working days of entitlement for an employee in their sixth (6^{th}) calendar year of service or more, except upon termination of service and subject to Article 12.02.

12.07 Vacation Carry Over

Up to one (1) year's vacation entitlement may be carried over to the next calendar year. Additional vacation entitlement may not be carried over to the next calendar year, except by authorization of the Department Head.

12.08 Payment of Vacation Pay

Vacation pay shall be paid not more than **seven (7)** working days prior to the commencement of the vacation period.

12.09 Illness or Death During Vacation

Sick leave or bereavement leave may be substituted for vacation when it can be established by the employee, to the satisfaction of the City, that a confining illness or an incapacitating accident or a death occurred while on vacation. The employee shall contact their immediate supervisor, as soon as possible, and a medical certificate must be obtained while on vacation and provided to the City upon the employee's return to work.12.10 Illness or Death During Vacation

ARTICLE 13 GENERAL HOLIDAYS

13.01 General Holidays

The City shall observe the following days as General Holidays with pay:

New Year's Day British Columbia Day

Family Day Labour Day

Good Friday Thanksgiving Day
Easter Monday Remembrance Day
Victoria Day Christmas Day
Canada Day Boxing Day

Truth and Reconciliation Day

and any other day declared or proclaimed as a general holiday by the Federal, Provincial, or Municipal Government.

(a) Sunday, Sunday Equivalent or Saturday

In the event that one of the within-named General Holidays falls on a Sunday, Sunday equivalent, it shall be observed the following working day. If the General Holiday falls on a Saturday, it shall be observed on that day, subject to the provision of Article 13.02.

(b) Newly Hired Employee

A newly hired employee shall be entitled to receive their regular pay for a General Holiday that falls within thirty (30) calendar days of commencement of employment, provided the employee has earned wages for at least fifteen (15) days prior to the General Holiday.

(c) Payment of Statutory Holidays - Casual Employees

An eligible employee working irregular hours on at least fifteen (15) of the thirty (30) days prior to the Statutory Holiday (eligibility based on being employed by the City for a minimum of thirty (30) calendar days) is entitled to an average day's pay for the Holiday. To calculate an average day's pay, divide the total wages earned in the thirty (30) day period (excluding overtime) by the number of days worked.

An eligible employee who has worked fewer than fifteen (15) days, but more than five (5), of the thirty (30) days prior to the Statutory Holiday is entitled to pro-rated Statutory Holiday pay. Pro-rated Statutory Holiday pay is calculated by dividing the total wages earned in the thirty (30) day period (excluding overtime) by fifteen (15).

An eligible employee who works on a Statutory Holiday will be paid time-and-one-half for the first eleven (11) hours worked and double time after eleven (11) hours. The employee will also be given the prorated amount for the Statutory Holiday (calculated based on the above formula).

13.02 Substituted Holiday

If by law or decree another day is substituted for the observance of any holidays listed in Article 13.01, the day of observance shall be considered as the holiday insofar as payment for the listed General Holiday is concerned.

13.03 Shift Work

Where an employee is on a shift that includes Saturday or Sunday as regular work days, then for the purposes of this section, the first (1st) scheduled day off shall be deemed to be Saturday, and the second (2nd) scheduled day off shall be deemed to be Sunday.

13.04 Payment for Work on General Holidays

An employee required to work on General Holidays shall receive the regular rate of pay for the day, plus overtime credits for actual hours worked, subject to the conditions of Article 10.

13.05 General Holiday While on Vacation

If a General Holiday falls within an employee's annual vacation period, the employee shall receive one (1) additional days' vacation with pay for such General Holiday(s).

13.06 Personal Day

After completion of the probation period (Article 8.01), an employee shall be entitled to one (1) day with pay each year as a **personal day**. Such **day** shall be arranged by mutual agreement between the employee and the Department Head. **This day may be taken in hourly increments and can be used for personal obligations.**

13.07 Temporary Assignment

An employee assigned by a temporary posting to a position paying a higher rate shall have General Holiday pay calculated at the higher rate provided the assignment continues through the General Holiday.

ARTICLE 14 HEALTH AND WELFARE

14.01 Benefit Plans

The City shall maintain the following benefits for Regular employees who have achieved seniority and work eighty (80) or more hours every four (4) weeks. Upon becoming eligible to participate in the Plans, an employee shall be enrolled either on the date they became eligible or no later than the first (1st) day of the next month, as determined by the carrier of each Plan.

Temporary employees working three (3) months or less from date of hire, shall receive fifteen and one quarter percent (15.25%) of their gross pay in lieu of benefits (sick leave, vacation, statutory holidays, medical, dental, extended health and life insurance). This can be extended with mutual agreement of the employer and the Union.

Temporary and Seasonal employees working longer than three (3) months will receive benefits like a regular employee.

Premium Sharing

(a) Medical Services Plan 100% City **Paid**

(b) Extended Health Plan 100% City **Paid**

Through the City's Plan Administrator or Carrier [includes **six hundred dollars (\$600.00)** vision care benefit every twenty-four (24) months per family member, one hundred dollars (\$100.00) towards eye exam every twenty-four (24) months per family member, Hearing Aid to **two thousand dollars (\$2,000.00)** every four (4) years]. One pair of Orthopedic shoes (no dollar limit) per year; and Orthotics two hundred dollars (\$200.00) every two (2) years.

To include an Extended Health Net drug plan

Paramedical services: **Eighty percent (80%)** reimbursement per visit **to an overall maximum of six hundred fifty dollars (\$650.00) per service per calendar year** for:

- 1. Chiropractor
- 2. Acupuncturists
- 3. Physiotherapist
- 4. Massage Therapy (service may include Massage, Rolfing, Reflexology)
- 5. Naturopath
- 6. Podiatrist

One hundred percent (100%) reimbursement per visit with an overall maximum of six hundred fifty dollars (\$650.00) per service per calendar year for:

- 1. Speech Therapist
- 2. Psychologist

(c) Medical Transportation

The City agrees to provide, through the City's Plan Administrator or Carrier, Medical Transportation for all regular employees. Premiums will be one hundred percent (100%) employer paid.

When ordered by the attending physician because in their opinion, treatment is not available locally, the following are included as eligible expenses for regular employees and their family members:

- 1. Transportation for a patient, by scheduled air, taxi to a maximum of thirty dollars (\$30.00) per claim, bus or rail to and from the nearest locale equipped to provide the required treatment within BC. This includes transportation for an attendant for the patient when required by a physician.
- 2. Where transportation has been provided above, accommodation will be covered for before and after treatment to a maximum of **one hundred and seventy five dollars (\$175.00)** per day for a total of seven (7) days. Accommodation is not provided for the attendant.
- 3. The cost of mileage incurred by a private automobile for the sole purpose of transporting to and from the nearest locale equipped to provide the required treatment.
 - The mileage payable will be **fifty-six cents (56¢)** per kilometre with distance calculated by the BC Ministry of Transportation Distance Calculator. *Mileage will not be reimbursable for medical travel within 200 kilometers*.
- 4. For all forms of transportation, the most practical and economical method of travel is to be taken.
- 5. Benefit amounts paid for all eligible medical travel claims will be reimbursed at one hundred percent (100%)
- (d) Dental Plan through the City's Plan Administrator or Carrier

100% City Paid

Basic Services benefits to one hundred percent (100%) reimbursement level

Major Services benefits to eighty percent (80%) reimbursement level

Orthodontics benefits to fifty percent (50%) reimbursement level to five thousand dollars (\$5,000.00) lifetime maximum per employee and each covered dependants.

(e) Group Life Insurance Plan including AD&D 50% City **Paid**(Insurance three (3) times annual salary) 50% Employee **Paid**Minimum hours' work - thirty (30) hours per week

- (f) An employee shall maintain enrolment in the benefit plans as permitted by the carrier, including when an employee's hours of work fall below the eligibility level for a period of two (2) months or less. When the reduced hours are as a result of being the successful applicant for a job posting this article does not apply.
- (g) An employee previously entitled to benefits with the City, must return to work for a period of longer than three (3) consecutive weeks to be re-signed onto City benefits.
- (h) The Employer shall notify the Union when the carrier changes.
- (i) Employees, who work beyond age sixty-five (65) and who were eligible for medical, dental and extended health coverage under Article 14.01, will continue to be eligible for such coverage, provided the City's Plan Administrator or Carrier permits coverage to continue and subject to the terms for such coverage established by the Carrier.
- (j) When an employee reaches age sixty-five (65) the Employee Life Insurance benefit will be reduced by fifty percent (50%) which is three (3) times annual basic earnings divided by two (2). Premiums are adjusted accordingly.

14.02 Pension (Municipal) Plan

The City agrees to participate in the Pension (Municipal) Plan. Regular fulltime employees shall enrol, and enrolment shall be effective upon completion of the probation period.

Regular part-time employees, working an average of one hundred and twenty (120) or more hours every four (4) weeks may choose to enrol in the Plan. Enrolment shall be effective upon completion of the probation period. Regular part-time employees, having two (2) years of service and earning thirty-five percent (35%) of the yearly maximum of Canada Pension Plan pensionable earnings for those two years may enrol in the Plan.

14.03 Sick Leave

An employee is not entitled to receive benefits under this section for injury or sickness while working for any other employer, or employee owned business, or while on layoff. The terms and provisions of sick leave benefits shall be as follows:

- (a) The employee must suffer an injury or illness not compensable under the Workers' Compensation Act, or injury not related to a motor vehicle accident for which ICBC wage loss benefits apply, subject to article 14.12.
- (b) The employee shall have achieved seniority.
- (c) Benefits shall commence on the first (1st) working day of absence due to accident, hospitalization (admission or referral to medical facilities),

- out-of-town medical referrals, and on the third (3rd) day due to sickness. All benefits shall be paid in accordance with this Agreement.
- (d) Benefits shall be one hundred percent (100%) of the employee's regular pay for a period of thirteen (13) weeks and shall be seventy-five percent (75%) of their regular pay for an additional thirteen (13) weeks. At the completion of the twenty-sixth (26th) week, all sick leave benefits shall cease.
- (e) In addition to Article 14.03 (c), an eligible employee is entitled to a benefit of six (6) days' paid leave per year at **one hundred percent** (100%) of their regular wages for absence due to sickness. These days are not cumulative.
- (f) When sickness occurs, the employee shall notify their immediate supervisor as soon as possible.
- (g) Benefits shall be payable at the regular pay periods as provided in this Agreement.
- (h) Where the City has concern regarding an employee's use of sick days, the City may require a medical certificate for any future sickness or injury by using the following procedure:
 - (i) the City shall meet with the employee and a Union executive representative to discuss the City's concerns.
 - (ii) the employee shall have the opportunity to respond to the City's concerns.
 - (iii) if the City and the Union are still concerned about the use of sick days, the City may require a medical certificate for each illness or injury commencing from the first day of absence.
 - (iv) the costs of supplying such a certificate shall be borne by the City when MSP does not cover the costs.
 - (v) this requirement may be in force for up to twelve (12) consecutive calendar months.
- (i) When an absence due to illness or accident exceeds three (3) working days, the employee shall, if requested, furnish the City with proof to its satisfaction that **they** were and/or is prevented from performing the duties of **their** position, and thereafter shall provide the same during the period of disability as and when requested by the City, and failure to do so shall result in the stopping of the payment of wages. Where a Doctor certifies that, prior to complete recovery, an employee is capable of returning to work to perform any light duties, the City may require the employee to return to work, provided such work is available.

The City has a right to a reasonable medical disclosure should it have a reasonable basis for seeking such. Medical disclosure will not simply be a matter of course. The accepted disclosure may include questions such as:

- (i) Has the employee been under the physicians care throughout the period of absence from work?
- (ii) Was the employee unfit to perform their regular job throughout the period of absence?
- (iii) Was the employee unfit to perform other suitable available employment during their absence?
- (iv) When will the employee be fit to return to their regular or other suitable employment?
- (v) Are there restrictions on the duties the employee can perform?
- (vi) Will the employee's condition be likely to cause further absenteeism in the future?
- (vii) Did the employee's illness or injury arise out of their employment (for WCB purposes)?

And any other legitimate and appropriate questions relating to the operation of the workplace and the job duties of the employees. Medical disclosure shall be in compliance with all regulatory standards, laws and guidelines.

- (j) Benefits are not payable under the following situations:
 - Disability resulting from wilfully self-inflicted injury or from an attempt at self-destruction;
 - (ii) Where the absence results from the use of drugs or alcohol, unless the employee is receiving continuing treatment under the care of a licensed physician;
 - (iii) Where the disability is caused by pregnancy, during the period commencing with the tenth (10th) week prior to the expected week of birth and ending with the sixth (6th) week after the week of the birth of the child.
 - (iv) Where the disability is resulting from work performed for any other employer or employee owned business.

(k) Recurring Disability

Should an employee who has received benefits due to a disability become disabled again, a new period of benefit shall only be established in the following circumstances:

- (i) If the new disability is totally unrelated to the previous disability and the employee has returned to work for at least one (1) day between disabilities;
- (ii) If the new disability is related to, or is a continuation of, the previous disability and the employee has returned to work for a period of at least two (2) weeks. In this circumstance, benefits shall commence on the first day of absence.

- (I) After an employee has been paid the maximum of twenty-six (26) weeks' pay provided for in Article 14.03 (d), the employee shall, if eligible, be placed on the LTD plan, without pay.
- (m) An eligible employee, while waiting for settlement of a disputed WCB or ICBC claim, shall be entitled to draw upon the sick leave benefits. Compensation subsequently received from WCB or ICBC shall be assigned to the City.

14.04 Pregnancy Leave

A female employee who requests leave under this section is entitled to up to seventeen (17) weeks of unpaid leave:

- (a) beginning:
 - (i) no earlier than thirteen (13) weeks before the expected birth date, and
 - (ii) no later than the actual birth date, and
- (b) ending:
 - (i) no earlier than six (6) weeks after the actual birth date, unless the employee requests a shorter period, and
 - (ii) no later than seventeen (17) weeks after the actual birth date.
 - An employee who requests leave under this section after the termination of a pregnancy is entitled to up to six (6) consecutive weeks of unpaid leave beginning on the date of the termination of the pregnancy.
 - 2) An employee is entitled to up to six (6) additional consecutive weeks of unpaid leave if, for reasons related to the birth or termination of the pregnancy, she is unable to return to work when her leave ends under subsection (1) or (2).
 - 3) A request for leave must:
 - (a) Be given in writing to the Employer,
 - (b) If the request is made during the pregnancy, be given to the Employer at least four (4) weeks before the day the employee proposes to begin a leave, and
 - (c) If required buy the Employer, be accompanied by a medical practitioner's certificate stating the expected or actual birth date or the date the pregnancy terminated or stating the reasons for requesting the additional leave under subsection (3).
 - (iii) The employee shall notify the City at least two (2) weeks prior to returning to the job.

A request for a shorter period under subsection (b) (i) must:

- (a) Be given in writing to the Employer at least two (2) weeks before the date the employee proposes to return to work, and
- (b) If required by the Employer, be accompanied by a medical practitioner's certificate stating the employee is able to resume work.

14.05 Continuation of Benefits

The City shall continue to provide the benefits of Article 14.01, 14.02 and 14.09 to an employee during the following absences, provided carrier eligibility requirements are met:

- (a) while in receipt of WCB or ICBC benefits, except where the employee has been laid off;
- (b) while in receipt of sick leave benefits, except where the employee has been laid off;
- (c) while on layoff, for one (1) additional calendar month beyond the month in which the layoff occurred;
- (d) while in receipt of LTD benefits, for a maximum of eighteen (18) months from date of disability;
- (e) while on unpaid leave of absence, up to five (5) consecutive work days;
- (f) in other circumstances, where an employee is eligible to return to work, an employee may continue coverage by arranging to pay the full premiums to the City, providing carrier eligibility requirements are met.
- (g) for one (1) additional calendar month for an employee who has retired.
- (h) **in** the event of the death of an employee, the City shall continue to pay its share of the monthly premiums for three (3) months to a medical plan, dental plan, and extended health plan for the benefit of the employee's dependant(s).

14.06 Long Term Disability

- (a) The City agrees to administer a Long Term Disability Plan for eligible employees.
 - The Plan and carrier shall be determined by the Union in consultation with the City.
- (b) All regular full-time employees, upon completion of the probation period, shall enrol in the Plan as a condition of employment. An eligible employee unless already covered by a wage loss replacement plan shall participate in the Long Term Disability Plan. Regular employees working less than full time with at least thirty (30) hours per week may enrol in the Plan on a voluntary basis and shall thereafter continue to participate so long as eligible.

- (c) The City agrees to deduct the premium from the earnings of each enrolled employee and forward the premiums and required reports to the carrier of the Plan, with a copy to the Union.
- (d) An employee shall continue to be covered by the Health and Welfare benefits of this Collective Agreement, as referred to in Article 14.05(d).
- (e) An employee while receiving benefits of the Plan or an eligible employee waiting for benefits to commence shall be considered to be on approved leave of absence.
- (f) The City agrees, that for all its employees and former employees enrolled in the Municipal Pension Plan, to advise the Municipal Pension Plan, in accordance with its procedures, of those persons drawing benefits under the Group Long Term Disability Plan.
- (g) An employee while receiving Long Term Disability will continue to accrue seniority but will not accrue vacation hours.

14.07 Leave for Birth or Adoption of Child

An employee shall be granted three (3) consecutive days off with pay for the birth **or the adoption** of **their** child **or children**.

14.08 Employee Assistance Program

The City and the Union agree to jointly administer an Employee Assistance Program. The premiums for the Program shall be **paid one hundred percent (100%)** by the City.

14.09 Adoption Leave

An employee, upon request, shall be entitled to the Parental Leave provisions of Article 14.10 when adopting a child.

14.10 Parental Leave

- 1) An employee who requests parental leave under this section is entitled to:
 - (a) For a parent who takes leave in relation to the birth of the child or children with respect to whom the parental leave is to be taken, up to sixty-one (61) consecutive weeks of unpaid leave beginning immediately after the end of the leave taken unless the Employer and employee agree otherwise.
 - (b) For a parent who does not take leave in relation to the birth of the child or children with respect to whom the parental leave is to be taken, up to sixty-two (62) consecutive weeks of unpaid leave beginning after the child's birth and within seventy-eight (78) weeks after that event;

- (c) For a birth father, up to sixty-two (62) consecutive weeks of unpaid leave beginning after the child's birth and within seventy-eight (78) weeks after that event, and
- (d) For an adopting parent, up to sixty-two (62) consecutive weeks beginning within seventy-eight (78) weeks after the child is placed with the parent.
- 2) If the child has a physical, psychological or emotional condition requiring an additional period of parental care, the employee is entitled to up to five (5) additional consecutive weeks of unpaid leave, beginning immediately after the end of the leave taken under subsection (1).
- 3) A request for leave must:
 - (a) Be given in writing to the Employer,
 - (b) If the request is for leave under subsection (1) (a) or (b), be given to the Employer at least four (4) weeks before the employee proposes to begin leave, and
 - (c) If required by the employee, be accompanied by a medical practitioner's certificate or other evidence of the employee's entitlement to leave.
- 4) An employee's combined entitlement to leave under and this section is limited to seventy-eight (78) weeks plus any additional leave the employee is entitled to under this section.

14.11 Sick Leave and ICBC or WCB

An employee who has been involved in a motor vehicle related accident and is injured and unable to work, shall not be eligible for sick leave benefits, however, the City will advance to the employee the equivalent amount of benefits as if they were eligible for sick leave benefits. When the employee receives wage loss benefits from ICBC these shall be reimbursed to the City in accordance with this Article. If a wage loss settlement from ICBC is not forthcoming, any monies paid to the employee by the City shall not require reimbursement.

An employee who receives wage loss benefits from ICBC or WCB shall reimburse the City for benefits received under Article 14.03 up to the amount of:

- (a) benefits received from the City under Article 14.03; or
- (b) benefits received from ICBC or WCB and designated as compensation for loss of wages, whichever is less.

14.12 General Leave of Absence

Leave of absence, without pay and without loss of seniority, may be granted by a Department Head to an employee requesting such leave, for good and sufficient cause. Such request shall be in writing, and the Department Head's reply shall be in writing and shall state the conditions of the employee's return to work. An employee should discuss continuation of benefits with the City to ascertain which benefits may be continued and the costs, if any, of continuing the benefits.

14.13 Medical Appointments

An employee, upon receiving approval of **their** immediate supervisor, shall be entitled to leave of absence for medical, dental and physiotherapy appointments for the employee and **their** spouse and children where such appointments cannot be reasonably scheduled outside of regular working hours. At a time mutually agreeable between the City and the employee, the employee shall have the opportunity to make up the lost time at their regular rate of pay. Such leave shall not be unreasonably withheld.

14.14 Family Illness

- (a) When no one in the family other than the employee can provide for the needs of an immediate member of the family during an illness, that employee may request an unpaid leave of absence to a maximum of five (5) days per calendar year. Immediate family **shall be those as defined in Article 15.02**. Such leave shall not be unreasonably withheld.
 - At a time mutually agreeable between the City and the employee, the employee shall have the opportunity to make up the lost time at **their** regular rate of pay.
- (b) The employee is eligible to use their sick leave days as defined in Article 14.03 (e) to offset their loss of wages due to caring for the child or immediate family members living in the household, when no one other than the employee is able to provide for their needs during an illness.

14.15 Employment Insurance Rebate

The City shall apply annually for the Employment Insurance Rebate **and advise the Union of the outcome**. The employees' share of the rebate shall be **dispersed accordingly** to the **employees**.

*Application to be made for any retroactive amounts the parties are eligible for.

14.16 Facility Passes

Employees and dependants shall be entitled to enrol in the City of Terrace Employee Discount Program, at a thirty-five percent (35%) discount.

ARTICLE 15 BEREAVEMENT LEAVE

15.01 Bereavement Leave

When death occurs to a member of the employee's immediate family, bereavement leave shall be granted to regular employees as follows:

- (a) One (1) regular work week with pay when **grieving or to attend** a funeral. Subject to approval of the Department Head, up to one (1) additional regular work week with pay when in attendance at a funeral if such additional time is required for travel.
- (b) Up to four (4) regular work weeks' unpaid leave of absence, subject to Department Head approval. Seniority and benefits will continue to accumulate during this leave.

15.02 Immediate Family

Members of the employee's immediate family are defined as their spouse (includes common-law), sons, daughters, mother, father, grandparents, grandchildren, brothers, sisters, legal guardians, sons-in-law, daughters-in-law, brothers-in-law, sisters-in-law, mother-in-law and father-in-law. This definition shall include step families.

15.03 Compensable Hours

Compensable hours under the terms of this Article shall be counted as hours worked for the purpose of computing vacation pay.

15.04 Pallbearer's and Mourner's Leave

An employee shall be granted leave of absence of one (1) day without loss of wages or benefits to attend a funeral or memorial service as a pallbearer. An employee shall be granted leave of absence for half ($\frac{1}{2}$) day without loss of wages or benefits to attend a funeral or memorial service as a mourner.

15.05 Compassionate Care Leave

Employees shall be eligible for leave without pay to care for a terminally ill family member in accordance with Bill 8 – 2006 Employment Standards (Compassionate Care Leave).

ARTICLE 16 CONTRACTING OUT

The Union would like to advise the City that, as of the signing of the new collective agreement commencing on January 1, 2022, we will be enforcing the language in the clauses in Article 16.

As soon as possible after the signing, the City and the Union agree to form a Joint Contracting Out Committee as per Article 16.03 and 16.04. When the City is contemplating hiring for work outside of the bargaining unit, they will provide notice to the Union as per

Article 16.02 and this committee will meet to ensure that the work meets the restrictions as outlined in Article 16.01.

16.01 Restrictions on Contracting Out

In order to provide job security for the members of the bargaining unit, the City agrees that all work or services normally performed by the employees shall not be subcontracted, transferred, leased, assigned or conveyed, subject to the following:

- (a) The employees are qualified to perform the work.
- (b) The equipment necessary to perform the work is available.
- (c) The work can be completed in the time available as governed by seasonal conditions.
- (d) The work can be performed by the City to the economic advantage of the City.

16.02 Advance Notice to Union

The Union shall be given advance notice of contracting out proposals for the purpose of discussing the method and work tasks of the contract. The City shall provide as much advance notice as possible and shall endeavour to provide at least thirty (30) days' notice.

16.03 Contracting Out Committee

The Union's contracting out committee shall have access to all necessary information which is not restricted pursuant to the *Freedom of Information and Privacy Act* to respond to any contracting out proposal. This includes information on final costs of contracted out work.

16.04 Joint Contracting Out Committee

The purpose of this Committee is to work together in good faith to discuss contracting out matters and to facilitate the requirements of Article 16.02 (Advance Notice to Union).

ARTICLE 17 REMUNERATION

17.01 Pay Schedules

Wage and salary rates shall be in accordance with Schedules A and B of this Agreement.

17.02 Allowances

In addition to those rates shown in Schedules A and B, the following differentials shall apply as applicable. Where differentials apply, an employee may accrue two (2) or more hours, but shall be paid no less than two (2) hours' differential pay per shift.

(a) Equipment:

- i. Manual Jack Hammer and Power Compactor Employees shall be paid **eighty-five (\$0.85)** per hour and only while operating a manual jack hammer or power compactor (jumping jack).
- ii. Chainsaw **One dollar and thirty cents (\$1.30)** per hour shall be paid to employees only while using chainsaw.

(b) Job Conditions:

- i. Grave Digging **Employees** shall be paid an additional **seventy cents** (**\$0.70**) per hour only while digging and backfilling graves.
- ii. Disinterment Employees shall be paid an additional five dollars (\$5.00) per hour only while digging and backfilling for the purposes of disinterment.
- iii. Dirty Pay Employees shall be paid an additional **two dollars** and five cents (\$2.05) only while required to manually clean and/or repair sanitary sewers while in use or required to clean up human feces or other human bodily fluids.
- iv. Charge Hand Employees shall be paid an additional **two dollars** and five cents (\$2.05) when designated as Charge Hands.

(c) Certifications:

- i. Mechanics possessing a CVSE Authorized Inspector Certificate **ninety cents (\$0.90)** per hour.
- ii. Class 5 Power Engineer or Refrigeration Operator Certification **ninety cents (\$0.90)** per hour.
- iii. EOCP Water Distribution Certification **ninety cents (\$0.90)** per hour, only when performing duties related to the water system.
- iv. EOCP Wastewater Certification **ninety cents (\$0.90)** per hour, only when performing duties related to the wastewater system.
- v. First Aid Level II A minimum of two (2) employees, designated by the Employer, from the Public Works building will hold a valid Occupational First Aid (OFA) Level II ticket, as approved by WorkSafe BC. Those holding the ticket will be compensated with a premium of ninety cents (\$0.90) per hour. Employees designated to hold a valid OFA Level II certificate will provide First Aid service when required. The Employer will keep records of the valid OFA Level II tickets and monitor expiration dates. Prior to expiration of the tickets, employees will be given the option to renew or let lapse the certification. Lapsed certifications will no longer receive the OFA Level

II premium. The Employer shall pay the costs associated with maintaining the certifications, plus expenses as provided for in Article 28.04 — Qualifications.

- (d) Shift Differentials the following differentials shall apply to regular positions. When an employee has their scheduled hours of work overlap two (2) shifts, the higher the premium shall apply to all hours worked when at least two (2) hours are worked in the higher premium shift:
 - i. Afternoon Shift 4:00 p.m. to 12:00 midnight **eighty-five cents (\$0.85)** per hour.
 - ii. Night Shift 12:00 midnight to 8:00 a.m. **one dollar and fifteen cents (\$1.15)** per hour.
 - iii. Early Morning Shift 5:00 a.m. to 1:00 p.m. **one dollar and fifteen cents (\$1.15)** per hour.

17.03 Temporary Assignment (Higher Paid Position)

An employee temporarily assigned to work in a higher paid position shall be paid the higher rate for all time worked, unless they are receiving training under active supervision. The training period shall not exceed three (3) weeks.

An employee being laid off or recalled is not considered to be temporarily reassigned for purposes of this Article.

17.04 Temporary Assignment (Lower Paid Position)

An employee temporarily assigned to work in a lower paid position shall maintain their regular rate of pay. An employee being laid off or recalled is not considered to be temporarily reassigned for purposes of this Article.

17.05 Jury or Court Witness Duty

The City shall grant leave of absence, with pay and without loss of benefits, to an employee who serves as a juror or a subpoenaed court witness. The City shall pay such an employee the difference between **their** normal earnings and that which the employee receives for such jury or court witness duties, excluding payment for travel, meals or other expenses. The employee shall present proof of service and of the amount of payment received for such jury or court witness duties.

17.06 Tool Allowance

Regular full-time mechanics who completed twelve hundred (1,200) hours in a calendar year shall be eligible for reimbursement of the cost of supplying or replacing work-related tools to a maximum of **eight hundred dollars** (\$800.00) per annum which can be accumulated from year to year, to a maximum banked amount of two thousand four hundred

dollars (\$2,400.00) at any one time. Receipts will be required by the City.

17.07 Boot Allowance

Regular full-time employees who have completed twelve hundred (1,200) hours in a calendar year shall have **one hundred percent (100%)** of the cost of safety toe work boots defrayed to a maximum of **four hundred dollars (\$400.00)** every two (2) years. Reimbursement shall be limited to those employees required by the City to wear safety toe boots.

** For interpretation's sake, an employee could claim for more than one (1) pair of boots during the two (2) year period, to a maximum **of four hundred dollars (\$400.00)** every two (2) years.

17.08 Day of Occupational Injury

The City shall continue to pay an injured employee their regular wages for the balance of the day of injury.

17.09 Recreation Department

An employee who is classified as a Recreation Leader or Lifeguard/Instructor and who is requested to perform maintenance duties shall be paid no less than the Labourer I rate of pay. Maintenance duties shall include, but not be limited to, painting and surface preparation for painting, annual locker maintenance, and annual lamp maintenance. Maintenance duties shall not mean routine janitorial and housekeeping duties.

17.10 Bathing Suits and Shoes

After one (1) year of employment, upon reaching fifty (50) hours of instruction time in a calendar year, employees required to wear bathing suits and indoor shoes in the performance of their jobs shall receive reimbursement of **one hundred percent (100%)** of the cost to a maximum of **two hundred fifty dollars (\$250.00)** every two (2) years.

** For interpretation's sake, an employee could claim for more than one (1) bathing suit during the two (2) year period, or a bathing suit and a pair_of shoes to a maximum of **two hundred fifty dollars (\$250.00)** every two (2) years.

17,11 RCMP Guards – Rest Periods

Terrace RCMP Guards will be compensated with an extra forty-five (45) minutes at straight time in lieu of breaks. Overtime, Shift Differential, and Holiday Pay will apply. All RCMP Guards will be required to remain in the cellblock for their 12-hour shift.

ARVICLE 18 PAYMENT SCHEDULE

18.01 Pay Days

Employees shall be paid on every other Friday.

18.02 Pay Period

Employees shall be paid all wages due up to the Friday midnight of the week preceding the Friday pay date, less statutory deductions.

18.03 Pay on Termination

Employees terminating shall be paid, wherever possible, on the day of termination or by the third working day following termination.

18.04 Automated Payroll Deposit

All employees shall be required to enrol in the City's direct deposit. For the purpose of correcting an error, or in the event of a layoff of an employee or for any other emergent issue, the City shall issue a manual cheque through the payroll computer system.

ARTICLE 19 JOB RECLASSIFICATIONS

19.01 Job Descriptions

As soon as reasonably possible, job descriptions shall be provided by the Employer upon request and reviewed on a continuing basis by the Union. All job descriptions shall be reviewed by the Union at least once every three (3) years. The job descriptions as presented shall become the recognized job descriptions. The Union shall be notified when a change is made to a job description or a new job description is developed. The new or changed job description shall not be in effect until two (2) weeks after the job description has been given to the Union.

19.02 New Positions

When a new position, not covered by Schedules A and B, is established during the term of this agreement, the following shall apply:

- (a) The City will develop the job description.
- (b) The position will be posted with an interim rate as "under review" according to Article 19.01.
- (c) After six (6) months the incumbent and the City will provide comments on the job description and the work performed.
- (d) The City and the Union shall negotiate a final rate of pay for the position.
- (e) Negotiations for the purposes of establishing new wage rates during the life of this Collective Agreement shall not require the opening of this Agreement.

- (f) Should the City and the Union not agree on a new rate for the position they may refer the matter to the Labour/Management Committee for review and recommendation.
- (g) Should agreement not be reached at the Labour/Management Committee, the matter may proceed under the Grievance procedure.

19.03 Changes in Classification

When any position not covered by Schedule "A" and "B" of this Agreement is established or the duties and responsibilities of a position are significantly changed during the term of this Agreement, the rate of pay shall be subject to negotiations between the City and the Union. If the parties are unable to agree on the reclassification and/or rate of pay of the job in question, such dispute shall be submitted to grievance and arbitration. The new rate shall become retroactive to the time the position was first filled by an employee.

An employee whose position or classification is changed to one with a lower salary, other than by their own volition, shall maintain the salary of their previous position but shall receive the rate for their new position when it equals or is greater than the rate for the previous position.

ARTICLE 20 GRIEVANCE PROCEDURE

20.01 Grievance Defined

Grievance means any difference which arises out of the interpretation, application, operation or any alleged violation of this Agreement, including any differences arising from the suspension or dismissal of any employee and including any question or difference as to whether any matter is arbitrable; such question or difference shall be finally and conclusively settled without stoppage of work in the manner herein contained.

20.02 Rights of Stewards

In order to provide an orderly and speedy procedure for the settling of grievances, the City acknowledges the rights and duties of the Union Grievance Committee and the Union Stewards.

The steward shall assist any employee whom the steward represents in presenting their grievance in accordance with the grievance procedure.

20.03 Rights of Grievor

The grievor shall have the right to be present at all meetings between the City and the Union when their difference is being discussed pursuant to the grievance procedure.

20.04 Replies in Writing

Replies and exchange of information to grievances shall be in writing at all stages which shall include the final disposition of the grievance.

20.05 Differences to be Discussed

Within fifteen (15) working days of the occurrence of an alleged difference, the employee(s) shall, with the Shop Steward in attendance or without if the employee(s) so chooses, discuss the matter with their immediate supervisor. If this discussion does not resolve the alleged difference, then it shall be processed as a grievance.

The Union is not prejudiced from filing a grievance at Step 1 of the grievance procedure of the Collective Agreement when the resolution to the above alleged difference adversely affects the Union.

20.06 Grievance Procedure

Step 1

If discussion with the immediate Supervisor does not produce a satisfactory settlement, then the employee(s) shall submit the dispute in writing to the designated management supervisor and with a Union representative in attendance, seek settlement of the dispute with the immediate supervisor within ten (10) working days of the discussion meeting with the same immediate supervisor.

The designated management supervisor shall meet with the employee(s) and the Union representative with a view to resolving the dispute within three (3) working days.

Step 2

If a satisfactory settlement is not reached after the dispute was submitted at Step 1, the dispute shall be submitted by the Union, in writing, within five (5) working days to the respective Department Head (or their authorized representative), who shall meet with the employee(s) and the Union with a view to resolving the dispute within five (5) working days.

Step 3

If a satisfactory settlement is not reached after the dispute was submitted at Step 2, the dispute shall be submitted, in writing, within five (5) working days to the Chief Administrative Officer, who shall meet with the employee(s) and the Union with a view to resolving the dispute within ten (10) working days. The Chief Administrative Officer may have the designated Step 2 Department Head present at the meeting as a resource person.

Step 4

If a satisfactory settlement is not reached after the dispute was submitted at Step 3, then the Union may submit the dispute to the City Council within ten (10) working days, who may meet with the Union with a view to resolving the dispute. The Union may, at the conclusion of Step 3, submit the dispute to arbitration rather than to the City Council.

Step 5

Failing satisfactory settlement after the dispute was submitted to the City Council or at Step 3, the Union may, within fifteen (15) working days, refer the matter to arbitration, with a single arbitrator to be selected by mutual agreement of the Union and the Employer.

20.07 General Grievance

Where a dispute involving a question of general application or interpretation of the Collective Agreement occurs, Steps 1 and 2 of the Grievance Procedure may be bypassed. The general grievance may be processed through the Grievance Procedure without the employee(s) in attendance, within fifteen (15) working days of the Union becoming aware of the difference.

20.08 City Grievance

- (a) In the event that the City wishes to file a dispute to grievance and arbitration, it shall first submit the matter in writing to the Union and seek settlement of the dispute.
- (b) If a satisfactory settlement is not reached within seven (7) working days after the matter has been submitted for settlement, the City may refer the matter to arbitration.

20.09 Grievance on Safety

- (a) An allegation, either by the Union or an affected employee, that certain work practices are unsafe to any person, may be grieved by the Union by filing the grievance with the Chief Administrative Officer. An employee may refuse to work under the unsafe conditions and shall not suffer a loss in pay. The City shall not reassign the alleged unsafe work while an investigation is taking place.
- (b) The appropriate Department Head and a Union representative shall inspect the site within one (1) hour of the occurrence of the incident causing the allegation of unsafe practice.
- (c) If a solution to remedy the work refusal cannot be agreed upon following the second investigation as per the Refusal of Work Procedure in the Health and Safety Program, both the Manager and the worker must notify Worksafe BC. A Prevention officer will then investigate the matter without undue delay and issue whatever orders are deemed necessary.
- (d) If satisfactory settlement cannot be reached after the Worksafe BC Investigation, the matter may be referred to arbitration.

20.10 Arbitration

(a) A grievance shall be heard by a single Arbitrator, the appointment of which shall be agreed to by both parties. If, after a reasonable period

- of time, the City and the Union are unable to agree on an Arbitrator, then either of them may apply to the Minister of Labour for the Province of British Columbia to appoint an Arbitrator.
- (b) The Board shall deliver its award in writing to each of the parties and the award of the majority of the Board shall be final and binding to the parties and shall be carried out forthwith.
- (c) Arbitration costs shall be apportioned in accordance with the provisions of Section 90 of the Labour Relations Code of British Columbia.

20.11 Amending Time Limits

The time limits mentioned in the Grievance and Arbitration Procedures may be extended by mutual agreement between the City and the Union.

20.12 Power to Amend

An arbitrator shall have the power to allow all necessary amendments to the grievance and the power to waive formal procedural irregularities in the processing of a grievance, in order to determine the real matter in dispute and to render a decision which **they deem** just and equitable.

ARTICLE 21 TERMINATION AND DISCIPLINE

21.01 Termination

An employee may be terminated for the following:

- (a) Just cause, with a copy of the notice of termination forwarded to the Secretary of the Union;
- (b) Absence without authorization or reasonable justification for more than two (2) consecutive working shifts. In this case, termination shall be effective at the end of the second (2nd) shift.

21.02 Discipline

When the City deems it necessary to discipline an employee, the City shall give written notice of such discipline to the employee with a copy to the Secretary of the Union.

21.03 Removal of Reprimand from Personnel File

The record of an employee shall not be used against them at any time when eighteen (18) months have elapsed since the issuance of a letter of reprimand, provided there has been no recurrence of a similar and/or other infraction.

Upon request of the employee, the reprimand shall be removed and destroyed once the eighteen (18) months have elapsed.

21.04 Right to Assistance from CUPE Representatives

An employee shall have the right to have Union representation present at any investigative discussion with supervisory personnel that the employee genuinely believes might be the basis of disciplinary action. Where the employer/supervisor intends to interview an employee for disciplinary purposes, the employer/supervisor shall so notify the employee in advance of the purpose of the interview in order that the employee may contact **their** representative to be present at the interview.

A steward or local Union officer shall have the right to consult with a CUPE staff representative or designate and to have **them** present at any discussion with supervisory personnel, which might be the basis of disciplinary action.

ARTICLE 22 LAYOFF AND RECALL

22.01 Layoff Guidelines

Pursuant to Article 2, Layoff Shall be Governed by the Following Guidelines:

- (a) Layoff shall be in reverse order of seniority with the employee with the least seniority being laid off first.
- (b) Laid off regular employees, who have not bumped another employee, shall have the right of recall to their former classification, should it become available. This shall mean the same position with the same number of hours in the position they held before being laid off.
 - They shall have the choice to accept recall, or not, to their former position if it is offered with less hours.
- (c) Recall shall be according to seniority for all casual and temporary work, provided the employee is capable (qualified) to do the work available. Other regular positions that become available will be posted according to the posting process and laid off employees shall have the right to bid on those jobs. The City shall not hire casual employees until regular employees on layoff, who are qualified to do the job, have been recalled.
- (d) Recall notices shall be issued by registered letter and shall be mailed to the last address given by the employee when the City has been unable to make personal contact with the employee.
- (e) An employee who has been recalled must notify the City of their intention to return to work within two (2) working days from the time the recall has been received by the employee. Employees on an approved leave of absence shall notify the City of their intention when their leave expires.
- (f) An employee must report to work within fifteen (15) calendar days from the date of their acceptance of recall, unless sick or injured, in which case the provisions of Article 22.05 (b) shall apply.

- (g) An employee who does not answer a recall notice in accordance with subsections (d) and (e) above, shall be deemed for all purposes to be self-terminated.
- (h) Notwithstanding Article 22.01 (e), an employee may refuse one (1) recall notice by notifying the City of their intention within one (1) working day from the time the employee received the recall notice. The right to refuse the recall shall only apply to a recall that is for work of less than two (2) weeks' duration.

22.02 Notice of Layoff

Other than casual or temporary employees, employees who are to be laid off shall be given notice in writing as specified below and the notice shall include their seniority and the seniority of all those junior to them. It is agreed that an employee may continue work on a day-to-day basis after the day of layoff stated in the notice and no further notice of layoff shall be required in such cases. The layoff notice shall state the effective date of the layoff. If work continues for a period of one (1) month beyond the specified layoff date, the layoff notice shall be deemed to be rescinded.

- (a) Where an employee has less than one (1) year of service one (1) week's notice;
- (b) Where an employee has one (1) year and up to three (3) years' service two (2) weeks' notice; and for each subsequent year of service, an additional week's notice, up to a maximum of eight (8) weeks' notice;
- (c) In the case of temporary layoff of less than two (2) months, one (1) week's notice.

If an employee has not had the opportunity to work the days as provided in this Article, **they** shall be paid for the days for which work was not made available.

In the case of a proposed temporary layoff of less than two (2) months that in fact exceeds two (2) months, the employee shall be entitled to full payment for the balance of notice that would have been required.

22.03 Bumping Procedure

- (a) The employee affected by a layoff shall be advised in writing that they may exercise their seniority and displace a less senior member of the bargaining unit in any position which they feel they are qualified to hold.
- (b) The employee wishing to exercise **their** seniority rights shall make written application to the Department Head within five (5) working days of receipt of the notice of layoff for employees with more than one (1) year of service, and within three (3) working days of receipt of the notice of layoff for employees with less than one (1) year service, advising of the position **they feel** qualified to hold, on the form provided.

- (c) The appropriate Department Head shall adjudicate the applicant's skills, qualifications and ability for the position and shall advise the employee, in writing, within ten (10) working days, of acceptance or rejection of the bump.
- (d) When an employee's application is accepted and it results in a less senior member of the bargaining unit being displaced, the displaced employee shall be given the same opportunity to exercise their seniority rights. Only after all applications to exercise seniority rights have been adjudicated, shall the displacements become effective.
- (e) When an employee is accepted to a position which carries a different classification, **they** shall be on trial in that new position for a period of sixty-five (65) working days. In the event that an employee proves unsuitable for the new position, they shall retain their seniority to apply for another position. At any time during the trial period the employee may choose severance or to revert to layoff with no recourse to further bumping.
- (f) Differences arising from layoffs shall be initiated in accordance with the grievance procedure.

22.04 Severance Pay Options

A regular employee who is laid off is entitled to choose severance pay at any time within thirty (30) calendar days from the effective date of layoff. Upon acceptance of severance pay, all rights under this Agreement are terminated. An employee on layoff and not recalled before loss of recall rights, shall automatically be paid severance pay within one (1) week of loss of recall rights.

- (a) Less than three (3) years' continuous service at the time of layoff: one week's current pay for each year of service, or major part thereof.
- (b) Three (3) or more continuous years' service at the time of layoff: for the first (1st) year of service, three (3) weeks' current pay; for the second (2nd) year of service, three (3) weeks' current pay; for each year thereafter, one-half (½) month's current pay.
- (c) Severance pay is limited to six (6) months' current pay.

22.05 Layoff While on Sick Leave or WCB

- (a) The City may, due to a reduction in work, lay off an employee who is absent from work due to illness or accident, provided the required notice of layoff has been given. Such an employee shall have the rights of a laid off employee, and the rights as provided in Article 22.
 - When the employee is pronounced fit to return to work the bumping rights outlined in Article 22.03 shall be exercised at that time.
- (b) A laid off employee, who is eligible for recall and is unable to return to work due to illness or accident, shall have their status changed to that

of being absent from work due to illness or accident, effective the date of recall, and shall enjoy all benefits of the Collective Agreement as if at work, except for payment of sick leave, wages and benefits. The City may request a medical certificate as provided in Article 14.03 (i). Upon recovery, the employee shall be returned to work.

(c) When an employee is laid off, while absent from work due to illness or accident, such layoff shall have no effect upon the employee's waiting period for long term disability plan benefits, providing the employee continues paying the full monthly LTD premium and subject to carrier eligibility requirements.

ARTICLE 23 EQUIPMENT CLASSIFICATION

23.01 Light Equipment

Self-propelled sidewalk sweepers, track-type snow removal equipment, Zamboni, tractors, five (5) yard dump trucks, hiab truck, line painter, Bomag pavement recycling plant, vibrating pavement roller, forklift and ride-on tractor mower.

23.02 Heavy Equipment

Grader, bulldozers, loaders one (1) yard and over, backhoes, dump trucks over five (5) yards, street sweepers and self-loading scraper.

23.03 Hiab Truck

The rate paid for the hiab truck shall be light equipment or heavy equipment based on the vehicle the hiab is attached to.

23.04 Special Equipment

Refuse packer with automatic transmission, vactor sump cleaner and hydraulic sewer cleaning equipment.

23.05 New Equipment

Classification of equipment not specifically mentioned in Articles 23.01 and 23.02 shall be negotiated prior to such equipment being introduced.

ARTICLE 24 PROTECTIVE CLOTHING

24.01 The City Shall Provide Protective Clothing and Safety Equipment in the Following Manner:

When authorized by a Department Head, clothing and equipment shall be furnished through central stores on a loan basis. The employee shall be required to sign for the item authorized and shall be responsible for the return of the items in good and serviceable condition, fair wear and tear excepted.

Only clothing and equipment listed herein shall be provided by the City.

(a) Identification

Items required for the purpose of special identification of City employees where required by Management.

(b) Protective Clothing

Hard hats, hipwaders, raingear, gloves, coveralls, slip-on steel toes (jack hammer) and such items as required by WCB.

(c) Safety Equipment

Safety goggles, painting masks, fluoride and chlorine masks, welding masks, ice cleats, respirators, first-aid kits and supplies, fire extinguishers, fluorescent traffic vests and control paddles, ear protectors and such items as required by WCB.

- (d) Leather aprons and surveyors' vests only to instrument persons as required.
- (e) Chaps and Skidoo suits as required.

24.02 All Other Items of Clothing

All other items of clothing including day-to-day clothing, safety toe and steel shank boots, rubber boots, winter clothing and all other similar items shall be the responsibility of the employees except as provided in Article 17.07.

ARTICLE 25 VACANCIES

25.01 Posting of Vacancies

- (a) Notice of all new positions and vacancies shall be posted on all employee notice boards and on the internal City website for five
 (5) working days before appointment is made. The name of the successful applicant(s) shall be posted on all employee notice boards.
- (b) An employee filling a temporary vacancy shall not subsequently be confirmed, by the City, as a regular appointment to that position without the vacancy first being posted as a regular position.
- (c) A part-time position which increases from less than twenty (20) hours per week to twenty (20) or more hours per week shall be posted. A position which increases from less than thirty (30) hours per week to thirty (30) or more hours per week shall be posted.
- (d) All notices shall contain the following information:

Nature of position, **minimum** qualifications, required knowledge and education, skills, shift, hours of work, wage rate. Such qualifications and requirements shall be those necessary to perform the job function.

All postings shall identify that it is a CUPE Local 2012 Bargaining Unit position and renumeration shall be according to the collective agreement negotiated between the parties.

25.02 Temporary Postings

Work of a temporary nature expected to be of more than four (4) weeks' duration shall be posted internally on employee notice boards **and on the internal City website**.

25.03 Acting Foreman

A temporary Acting Foreman may be appointed at ninety-five percent (95%) of the appropriate Working Foreman rate. A Temporary Acting Foreman may be paid, as above, for up to sixty-five (65) consecutive days of work, and thereafter, or sooner if deemed appropriate by the City, shall be paid the rate for the appropriate Working Foreman classification.

25.04 Temporary Assignments

Where regular employees accept work in a temporary assignment, they shall return to their own regular positions after the temporary assignment is over. Termination of the temporary assignment shall not be considered lay off.

ARTICLE 26 TECHNOLOGICAL CHANGE

26.01 Advance Notice

Three (3) months before the introduction of any technological change which affects the rights of employees, conditions of employment, wage rates or workloads, the City shall notify the Union of the proposed change.

26.02 Changes

Any such change shall be made only after the Union and the City have reached agreement on such change.

26.03 No Agreement on Results of the Change

If the City and the Union fail to agree on the results of the change, the matter shall be subject to Arbitration, pursuant to Article 20.

26.04 Regular Employee

Pursuant to Article 26.06 (b), no regular employee shall be dismissed by the City because of technological changes, subject to Articles 26.05 and 26.06.

26.05 Training Period for Technological Changes

In the event that the City should introduce technological changes which require new or greater skills than are possessed by employees under the present method of operation, such employees shall at the expense of the City be given a period of time, not to exceed one (1) year, during which they may perfect or acquire the skills necessitated by that change. There shall be no reduction in wage or salary rates during the training period of any such employee and no reduction in pay upon being reclassified in the new position requiring the use of the new skills.

26.06 Displaced Employees

- (a) In the case where an employee is displaced from **their** job by virtue of technological change, an equitable attempt shall be made by the City to find the employee agreeable employment elsewhere in the City's operation at a rate of pay that exists for the job.
- (b) Where agreeable employment cannot be arranged, employment with the City shall be terminated and one (1) week's severance pay for each year of service with the City shall be paid to the employee when termination results due to technological change.

ARTICLE 27 UNION DUTY

27.01 Union Leaves With Pay

Union representatives shall be granted permission from the Department Head or their immediate supervisor to be absent as hereinafter mentioned and shall suffer no loss of pay under the following circumstances:

- (a) **Four (4)** representatives for the purpose of collective bargaining;
- (b) For the purpose of grievance procedures pursuant to Article 20;
- (c) Three (3) representatives to participate on a Labour/Management Committee.
- (d) Representation on the City's Occupational Health and Safety Committee, or other joint Committees between the City and the Union.

27.02 No Overtime on Union Leaves with Pay

Under no circumstances shall overtime be paid for or accumulated as a result of absence pursuant to the provisions of the preceding clause.

Payment for approved absences shall be paid for at the representative's basic hourly rate at the time of such absence.

27.03 Union Leaves Without Pay

- (a) Applications for leave of absence for Union duty shall not be unreasonably withheld.
- (b) Upon application to, and upon receiving the permission of the Chief Administrative Officer, representatives of the Union shall be granted leave of absence, without pay, for the purpose of attending educational seminars, meetings and conventions of CUPE, the BC Federation of Labour and/or the Kitimat-Terrace & District Labour Council.
- (c) An employee elected or appointed to any of its committees or elected or appointed as an officer of CUPE, it's affiliated or chartered bodies and any labour organizations with which the Union is affiliated, may be allowed leave of absence without pay but without loss of benefits to attend their duties. Such an employee shall inform the City in writing

- of their election or appointment. Such leave shall not be unreasonably withheld.
- (d) With respect to leave of absence granted under subsections (b) and (c) above, the City shall continue to pay such representatives their regular wage or salary and thirty percent (30%) for benefits, and shall render an account to the Union for that amount. The Union shall reimburse the City within thirty (30) days.
- (e) Upon application to and receiving the permission of the Chief Administrative Officer, an employee shall be granted leave of absence without pay and benefits, but without loss of seniority, so that the employee may be a candidate in Federal or Provincial elections.
- (f) Upon application to, and upon receiving the permission of the Chief Administrative Officer, an employee who is elected pursuant to subsection (e) above, shall be granted a leave of absence without pay and benefits, but without loss of seniority, during their term of office.
- (g) An employee who is elected or selected for a full-time position with the Union, or any body with which the Union is affiliated shall, upon application to and upon receiving the permission of the Chief Administrative Officer, be granted a leave of absence without pay and benefits, but without loss of seniority for a period of up to **two (2)** years. Upon application to, and upon receiving approval of the Chief Administrative Officer, such leave of absence shall be reviewed every **two (2) years**, during the employee's term of office.
- (h) The Union shall be allocated twenty (20) working days a year with pay for its members to attend educational courses on occupational health and safety. Payment shall be at the employee's regular rate of pay. Application for leave shall be made to and approved by the Chief Administrative Officer. The Union shall briefly state the training course content when seeking this paid leave.

Such leave shall not be unreasonably withheld.

ARTICLE 28 GENERAL CONDITIONS

28.01 Changes to Agreement

With the exception of Schedules A and B, any changes deemed necessary in this Agreement may be made by mutual consent at any time during the term of this Agreement. Such changes shall be made in the form of a "Memorandum of Agreement" attached to this Agreement.

28.02 Printing of Agreement

The City and the Union agree to share equally in the cost to print the Agreement in booklet form.

28.03 Loss of Employee Qualifications

- (a) When an employee loses a qualification and as a result is unable to perform the job **they were** hired to perform, the City shall, where possible, assign the employee to alternate work, provided work is available and the employee possesses the necessary skills and ability to perform the work.
- (b) An employee shall not be removed from their position, as a result of being disqualified to operate a motor vehicle, when the major responsibilities of their job can be discharged without unreasonable delay and provided there is no increased cost to the City, or when another employee can temporarily be assigned the driving tasks, where equipment or vehicle operation is not the major responsibility of the position and there is no increased cost to the City.
- (c) An employee assigned to another position shall receive the rate of pay for that new position.
- (d) When alternate work is not available the employee shall be laid off and then acquire normal recall rights.
- (e) An employee being laid off under the provisions of this Article does not require any notice of layoff.
- (f) When an employee again qualifies for their original position, **they** shall be reinstated, provided **they** still **have** retained seniority.

28.04 Qualifications

An employee who is required to take a course or written examination to renew or maintain qualifications requisite to **their** position, shall, upon successful completion, be reimbursed the course or examination fees and travel costs as per the City of Terrace Policy #13. Such an employee shall not suffer loss of wages or benefits while attending such a course or examination.

The City shall not pay wages for attendance at such course or written examination held outside the employee's regular hours of work. Attendance at such courses is subject to Department Head approval.

28.05 Workers' Compensation Pension

When a full pension is granted by the Workers' Compensation Board, the employee shall be terminated, effective the commencement date of the pension.

28.06 Day of Mourning

The City and the Union agree to recognize April 28^{th} as the National Day of Mourning to remember those workers injured and killed on the job. On April 28^{th} at 11:00 a.m., there shall be a cessation of work for one (1) minute in memory of those workers.

28.07 In-Service Training (Aquatic Staff)

The City, the Union and the employees recognize the importance and advantages of in-service training in the aquatic field.

Paid in-service training sessions will be scheduled at least four (4) times annually. All in-service training sessions will be compulsory, unless a reasonable excuse is given.

The paid in-service training sessions for the regular full-time employees, if taken outside the normal schedule, will accrue as overtime and may be taken as per Article 10.

28.08 Minimum Staffing

During the times when the pool is open for public swimming, the minimum staffing level shall be two (2) guards.

28.09 Maximum Deck Time

The City recognizes the safety factor related to minimizing deck time duty for employees, and will make every effort to limit deck time under usual circumstances to less than one (1) hour and not exceeding two (2) hours.

ARTICLE 29 TERM OF AGREEMENT

29.01 Term of Agreement

This Agreement shall take effect January 1, 2022, and shall remain in effect until December 31, 2026, and thereafter from year to year unless written Notice of Intent to amend the Agreement is given by either party to the other party in accordance with the provisions of the Labour Relations Code of British Columbia.

IN WITNESS WHEREOF, the City and the Union by their authorized representatives, have affixed their signatures hereto on the _______, 2022.

THE CITY OF TERRACE:	CANADIAN UNION OF PUBLIC EMPLOYEES, LOCAL 2012:			
Tage appealably professioners	Charmel			
HR MANAGER	PRESIDENT			
#3				
CHIEF ADMINISTRATIVE OFFICER	NEGOTIATING COMMITTEE			

SCHEDULE "A" - WAGES

Schedule A Wages	1-Jan- 22	1-Jan- 23	1-Jan- 24	1-Jan- 25	1-Jan- 26
Position	3.50%	3.00%	3.00%	2.50%	2.50%
Recreation Leader (Pool & Arena)	21.56	22.21	22.87	23.44	24.03
Animal Control Facility Attendant I	24.80	25.54	26.31	26.97	27.64
Labourer I (Student Rate)	24.80	25.54	26.31	26.97	27.64
Lifeguard/Instructor I	25.99	26.77	27.57	28.26	28.97
Animal Control Facility Attendant II	27.55	28.38	29.23	29.96	30.71
*Gardener/Landscaper (Seasonal)	27.66	28.49	29.35	30.08	30.83
Lifeguard/Instructor II	27.66	28.49	29.35	30.08	30.83
Guard	29.94	30.84	31.76	32.56	33.37
Labourer I	29.94	30.84	31.76	32.56	33.37
Recreation Attendant I	29.94	30.84	31.76	32.56	33.37
Utility Person In-Training	29.94	30.84	31.76	32.56	33.37
Animal Control Officer	30.58	31.50	32.44	33.25	34.08
Construction Maintenance Attendant I	30.58	31.50	32.44	33.25	34.08
Lifeguard/Instructor III	30.58	31.50	32.44	33.25	34.08
Labourer II	30.58	31.50	32.44	33.25	34.08
Labourer III	31.54	32.49	33.46	34.30	35.16
Parks, Trails & Greenspace Operator	31.54	32.49	33.46	34.30	35.16
Recreation Attendant II	31.54	32.49	33.46	34.30	35.16
Engineering Tech I	32.20	33.17	34.16	35.02	35.90
Environmental Services Tech I	32.20	33.17	34.16	35.02	35.90
Light Equipment Operator	32.20	33.17	34.16	35.02	35.90
Warehouse Person	32.20	33.17	34.16	35.02	35.90
Utility Person	32.98	33.97	34.99	35.86	36.76
Construction Maintenance Attendant II	33.75	34.76	35.81	36.70	37.62
Downtown Safety Officer	33.75	34.76	35.81	36.70	37.62
Engineering Tech II	33.75	34.76	35.81	36.70	37.62
Head Lifeguard/Instructor	33.75	34.76	35.81	36.70	37.62
Heavy Equipment Operator I (Trainee Position)	33.75	34.76	35.81	36.70	37.62
Mechanic I	33.75	34.76	35.81	36.70	37.62
Purchasing Agent I	33.75	34.76	35.81	36.70	37.62
Environmental Services Tech II	34.32	35.35	36.41	37.32	38.25
Heavy Equipment Operator II	34.98	36.03	37.11	38.04	38.99
Special Equipment Operator	34.98	36.03	37.11	38.04	38.99
*Aquatic Facility Maintenance Technician	35.61	36.68	37.78	38.72	39.69
Bylaw Compliance Officer I	35.61	36.68	37.78	38.72	39.69
Construction Maintenance Attendant III	35.61	36.68	37.78	38.72	39.69
Parks Supervisor	35.61	36.68	37.78	38.72	39.69
Purchasing Agent II	35.61	36.68	37.78	38.72	39.69
Recreation Attendant III	35.61	36.68	37.78	38.72	39.69
Environmental Services Tech III	35.61	36.68	37.78	38.72	39.69
*Aquatic Supervisor	36.97	38.08	39.22	40.20	41.21

Mechanic II	36.97	38.08	39.22	40.20	41.21
Building/Licence Official I	38.93	40.10	41.30	42.33	43.39
Bylaw Compliance Officer II	38.93	40.10	41.30	42.33	43.39
Design Engineer/Technologist	38.93	40.10	41.30	42.33	43.39
Heavy Equipment Operator III	38.93	40.10	41.30	42.33	43.39
Aquatic Foreman	40.97	42.20	43.47	44.55	45.66
Building/Licence Official II	40.97	42.20	43.47	44.55	45.66
Working Foreman - Building Maintenance	40.97	42.20	43.47	44.55	45.66
Working Foreman - Construction	40.97	42.20	43.47	44.55	45.66
Working Foreman - Electrical	40.97	42.20	43.47	44.55	45.66
Working Foreman - Environmental Health	40.97	42.20	43.47	44.55	45.66
Working Foreman - Parks & Facilities	40.97	42.20	43.47	44.55	45.66
Working Foreman - Roads & Drainage	40.97	42.20	43.47	44.55	45.66
Working Foreman - Shop	40.97	42.20	43.47	44.55	45.66
Working Foreman - Water & Waste	40.97	42.20	43.47	44.55	45.66
Building/Licence Official III	43.95	45.27	46.63	47.79	48.98
Engineering Supervisor	43.95	45.27	46.63	47.79	48.98
Plant Foreman	43.95	45.27	46.63	47.79	48.98
Road Foreman	43.95	45.27	46.63	47.79	48.98
*These wages are subject to Article 19.02, as they are still under review at the time of bargaining.					

All wages shall be retroactive to January 1, 2022.

SCHEDULE "B" - WAGES

Schedule B Wages	1-Jan- 22	1-Jan- 23	1-Jan- 24	1-Jan- 25	1-Jan-26
Position Schedule B Wages	3.50%	3.00%	3.00%	2.50%	2.50%
Day Camp Leader	17.72	18.25	18.80	19.27	19.75
Day Camp Supervisor	21.56	22.21	22.87	23.44	24.03
Recreation Programmer	28.58	29.44	30.32	31.08	31.86
Accounting Clerk I	30.70	31.62	32.57	33.38	34.21
Planner I	30.72	31.64	32.59	33.41	34.25
Accounting Clerk II	31.54	32.49	33.46	34.30	35.16
Accounting Clerk II (Aquatic Clerk)	31.54	32.49	33.46	34.30	35.16
Accounting Clerk II (Clerk Receptionist)	31.54	32.49	33.46	34.30	35.16
Development Services/Public Works Clerk	31.54	32.49	33.46	34.30	35.16
Secretary I (Admin)	31.54	32.49	33.46	34.30	35.16
Accounting Clerk III	32.26	33.23	34.23	35.08	35.96
Accounting Clerk III (Accounts Payable)	32.26	33.23	34.23	35.08	35.96
Accounting Clerk III (Accounts rayable) Accounting Clerk III (Building & Business Licence Clerk)	32.26	33.23	34.23	35.08	35.96
Accounting Clerk III (Building & Business Elcence Clerk) Accounting Clerk III (Finance Support Clerk)	32.26	33.23	34.23	35.08	35.96
Accounting Clerk III (Finance Support Clerk) Accounting Clerk III (Leisure Services)	32.26	33.23	34.23	35.08	35.96
Accounting Clerk III (Public Works)	32.26	33.23	34.23	35.08	35.96
Accounting Clerk III (Fublic Works) Accounting Clerk III Finance Service Coordinator	32.26	33.23			
Secretary II (CIPC)	32.26	33.23	34.23	35.08	35.96
Accounting Clerk IV		33.23	34.23	35.08 35.79	35.96
Accounting Clerk IV (Cashier)	32.91 32.91	33.90	34.91 34.91	35.79	36.68 36.68
Accounting Clerk IV (Cashler) Accounting Clerk IV Finance Clerk	32.91	33.90	34.91	35.79	36.68
Departmental Services Clerk I		33.90	34.91	35.79	
Detachment Clerk (RCMP)	32.91 32.91	33.90	34.91	35.79	36.68 36.68
Accounting Clerk V	33.60	34.61 34.61	35.65	36.54	37.45
Accounting Clerk V (Finance Clerk) Accounting Clerk V (Payroll)	33.60		35.65	36.54 36.54	37.45
CAD/GIS Technician I	33.60	34.61	35.65		37.45
Departmental Services Clerk II	33.60 33.60	34.61 34.61	35.65 35.65	36.54 36.54	37.45 37.45
Programs Supervisor		34.76	35.81		37.43
Accounting Clerk VI Payroll & Benefits Administrator	33.75	35.71		36.70	38.64
	34.67		36.78	37.70	
Purchasing Supervisor Public Works Records Coordinator - RCMP	34.67	35.71	36.78	37.70	38.64
	35.22	36.28	37.37	38.30	39.26
CAD/GIS Technician II	36.37	37.46	38.59	39.55	40.54
Planner II	37.19	38.31	39.46	40.44	41.45
CAD/GIS Technician III	38.13	39.27	40.45	41.46	42.50
Mapping/Planning Technician	38.93	40.10	41.30	42.33	43.39
Planning Technician	41.37	42.61	43.89	44.99	46.11
Planner III	42.04	43.30	44.60	45.72	46.86

between

CITY OF TERRACE

and

CUPE LOCAL 2012

Re: Management Staff on LTD Plan

The Union agrees to permit the enrolment of the regular management staff of the City in its Long Term Disability Plan on the same basis as Union members may enrol.

Signed this 5th day of October, 2022	2.
THE CITY OF TERRACE:	CANADIAN UNION OF PUBLIC EMPLOYEES, LOCAL 2012:
nggan	Ufamal
HR MANAGER	PRESIDENT
_KB	
CHIEF ADMINISTRATIVE OFFICER	NEGOVIATING COMMITTEE

Between

CITY OF TERRACE

and

CUPE LOCAL 2012

Re: Change of Shift Public Works Department - Schedule A Employees

Under special circumstances, for the duration of this collective agreement, the parties may agree to change shifts to hours other than those set out in Article 9. The Union shall not unreasonably withhold agreement if the following criteria are met:

- 1. The shift change shall be for a specified duration (number of days or duration of job).
- 2. An employee shall not lose any shift premium(s) to which **they** would have been entitled if the shift change had not taken place.
- 3. An employee shall be entitled to any shift premium(s) generated by the new shift, but shall not be pyramided with (2) above.
- 4. All other articles of the collective agreement apply.

Between

CITY OF TERRACE

and

CUPE LOCAL 2012

Re: Article 17.02(d)

For the duration of this collective agreement, the parties agree that shift differential shall apply to regular full time, regular part time, temporary, and seasonal employees as well as **Casual RCMP Guards when scheduled for a twelve-hour shift** under Article 17.02 (d). All other articles of the collective agreement apply.

Signed this 5th day of October, 2022.	
THE CITY OF TERRACE:	CANADIAN UNION OF PUBLIC EMPLOYEES, LOCAL 2012:
HR MANAGER	PRESIDENT
CHIEF ADMINISTRATIVE OFFICER	NEGOTIATING COMMITTEE

Between

CITY OF TERRACE ("The City")

And

CANADIAN UNION OF PUBLIC EMPLOYEES, LOCAL 2012 ("The Union")

Re: Wage Review for Red Seal Ticketed Staff and Article 17.02 Allowances

- 1. The parties agree that, as the sole bargaining agent for its members, all changes to the collective agreement must be negotiated exclusively with the union and not with any individual or groups of its members.
- 2. The City and the Union recognize that there are significant challenges with recruitment and/or retention of qualified Red Seal ticketed staff. In response to those challenges, the City conducted a wage review for staff that hold tickets. Although the Union was not involved in the initial process, the parties now agree that:
 - i. Effective August 8, 2022 the hourly wage rates for the following positions will be adjusted as follows, and will include Article 17.02 allowances which were previously paid in addition to the hourly wage:

CLASSIFICATION	CURRENT WAGE RATE	NEW WAGE RATE
Aquatic Facility Maintenance Technician	\$35.61	\$36.97
Construction Maintenance III	\$35.61	\$36.97
Mechanic II	\$36.97	\$40.97
Working Foreman – Building Maintenance	\$40.97	\$43.95
Working Foreman – Electrical	\$40.97	\$43.95
Working Foreman – Environmental Health	\$40.97	\$43.95
Working Foreman – Parks & Facilities	\$40.97	\$43.95
Working Foreman – Shop	\$40.97	\$43.95
Working Foreman – Roads & Drainage	\$40.97	\$43.95

^{*}Road Foreman is currently at \$43.95 per hour and the parties acknowledge that this rate shall remain and includes the allowance that was previously paid in addition to the hourly wage.

COLLECTIVE AGREEMENT: CITY OF TERRACE AND CUPE LOCAL 2012 January 1, 2022 – December 31, 2026

- 3. The City and the Union also recognize that there are classifications that are receiving Article 17.02 allowances in addition to their hourly wage for certifications which are required in order to hold the position. Both parties see the value in building these allowances into the hourly wage, and so agree that:
 - i. Effective August 8, 2022, the hourly wage rates for the following positions will be adjusted to include Article 17.02 allowances which were previously paid in addition to the hourly wage:

CLASSIFICATION	CURRENT WAGE RATE	NEW WAGE RATE (With built-in allowance(s))
Environmental Technician II	\$34.32	\$35.22
Environmental Technician III	\$35.61	\$36.51
Recreation Attendant II	\$31.54	\$32.44
Recreation Attendant III	\$35.61	\$36.51
Recreation Attendant IV Chief Engineer	\$36.97*	\$37.87*

^{*} Recreation Attendant IV Chief Engineer is a new position at the signing of this letter and is under review for a six-month period.

- 4. The wage adjustments will apply to all current and new incumbents and will be treated as pensionable earnings.
- 5. The parties agree that if the job description requires that the ticket(s) must be held in order to perform the duties of the job ("job requirements") then the allowance will be built into the new wage rate, above.
- 6. Article 17.02 allowances will remain in the Collective Agreement, with the exception of the CVSE Authorized Inspector Certificate and the Class 5 Power Engineer or Refrigeration Operator Certification.
- 7. Remaining Article 17.02 allowances may be claimed when an employee holds the certification that is not a job requirement for their classification and performs work which requires that certification. Allowance(s) will only be paid for those hours performing duties which require the certification.
- 8. The parties recognize that recruitment and retention challenges with specific bargaining unit positions may occur over the life of the collective agreement and so the parties agree that prior to any further wage reviews, a joint committee consisting of two representatives from each of the Union and the Employer will be formed.
 - (a) Should an employee, the Union, and/or the City feel a position is improperly classified, believes the duties and/or responsibilities of a position have significantly changed, or believes that a wage review is justified for retention and/or recruitment issues, they shall submit a completed "Request for Reclassification/Wage Review" form to the

- Human Resources Department. The form must be accompanied by all the required documentation as laid out on the form.
- (b) Upon receipt of the completed form, the Human Resources Department will arrange any necessary meetings between the committee members. Any member of this committee cannot have any monetary interest in the outcome of the adjudication. Every effort will be made by both parties to complete the adjudication process in a timely manner.
- (c) Once the committee agrees, the City and the Union agree to update Schedules A & B of the Collective Agreement reflecting any new hourly wage rates. Any adjustment in wages arising from the adjudication will be retroactive to the date that the properly completed "Request for Reclassification/Wage Review" form was received by the Human Resources Department.
- (d) An employee whose position or classification is changed to one with a lower wage rate shall maintain their previous wage rate until the new wage rate is equal to or greater than the previous rate for the position or classification (red circled).
- (e) Should the City and the Union be unable to come to an agreement, the matter may proceed under the Grievance procedure.
- 9. The Union reserves its rights to file any future grievances and/or reclassification requests of the above noted positions or any other bargaining unit position.

CHANGES IN CLASSIFICATION/REQUESTS FOR WAGE REVIEW INFORMATION FORM

Em p	oloyee Name:
Curr	rent Position:
1.	Please provide a list of the differences in the duties and responsibilities noted in the job description in comparison with the actual duties and responsibilities performed currently in the position (attach additional information if necessary):
2.	Please attach a comparison of positions within the City (if any) and three (3) positions from other municipalities that have comparable duties and responsibilities to the position under review. If there are no comparable positions within the City, then a minimum of five (5) positions from other municipalities is required. (Note: include job descriptions and pay rate for all positions).
3.	Please provide information on any new skills or qualifications that are required to do the changed job.
4.	Please indicate the wage rate you are seeking for the position:
s y Series (1968	
Date	Complete Package Received by Personnel Department: